

G OPEN ACCESS

Citation: Li X (2020) The effectiveness of internal control and innovation performance: An intermediary effect based on corporate social responsibility. PLoS ONE 15(6): e0234506. https://doi.org/10.1371/journal.pone.0234506

Editor: Stefan Cristian Gherghina, The Bucharest University of Economic Studies, ROMANIA

Received: January 22, 2020

Accepted: May 26, 2020

Published: June 11, 2020

Copyright: © 2020 Xiao Li. This is an open access article distributed under the terms of the <u>Creative</u> Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: The soft science research program of Henan province, China (No. 202400410409).

Competing interests: The author has declared that no competing interests exist.

RESEARCH ARTICLE

The effectiveness of internal control and innovation performance: An intermediary effect based on corporate social responsibility

Xiao Li₀^{1,2}*

1 Economics and Management School, Zhongyuan University of Technology, Zhengzhou, Henan, China,

2 Economics and Management School, Wuhan University, Wuhan, Hubei, China

* xiaoli78@aliyun.com

Abstract

From the perspective of the effectiveness of internal control, this study analyzes the influence of internal control on innovation performance and internal control on corporate social responsibility (CSR), and then analyzes the intermediary effect of CSR between internal control and innovation performance. The results show that the improvement of the effectiveness of internal control has a significant promoting effect on innovation performance, and promotes enterprises to strengthen CSR performance. Meanwhile, CSR activities take a significant intermediary effect in the process of improving innovation performance through internal control. Finally, it is suggested that state-owned enterprises and non-state-owned enterprises should communicate and cooperate, strengthen the construction of internal control system, and improve innovation performance and CSR practices. Furthermore, the intermediary effect of CSR activities in the process of improving innovation performance through internal control should be brought into play, so as to return the expectations and demands of stakeholders.

1. Introduction

Internal control and technological innovation are of great significance to the survival and development of enterprises. And the internal control has become an important system for Chinese enterprises to improve their social responsibilities [1]. Moreover, enterprises with strategic CSR achieve growth through both their product and their process innovations [2]. Technological innovation is the strategic support for building a modern economic system. Therefore, from the perspective of the effectiveness of internal control, it is of practical significance to explore the joint mechanism of internal control and CSR on innovation performance. From 2008 to 2010, Chinese five ministries and commissions, including the Ministry of Finance, successively issued the "Basic Norms for Enterprise Internal Control" and the "Supporting Guidelines for Enterprise Internal Control", promoting the development of the construction of internal control. Where, the "Supporting Guidelines for Enterprise Internal control", the "Evaluation Guidelines



for Internal Control" and the "Audit Guidelines for Internal Control". These normative documents clearly point out that internal control is a process carried out by the board of directors, board of supervisors, managers and all employees to achieve the control objectives. The elements of internal control include the internal environment, risk assessment, control activities, information and communication, and internal supervision. As an institutional arrangement, internal control aims at reasonably ensuring the legal operation and compliance, safety of assets, and reliability of financial information, improving the efficiency and effect of operations, and thus promoting the realization of the development strategy of enterprises. Since the promulgation of the current standard system of internal control, Chinese listed enterprises have achieved some outstanding results in the implementation, evaluation and information disclosure of internal control, providing a good reference for the system construction of unlisted enterprises, and building a good micro foundation for the steady operation of the market economy. The improvement of the effectiveness of internal control can enhance the sustainable development of enterprises to a certain extent [3]. The effective internal control is conducive to improving the financing efficiency [4], and significantly and positively affects the investment efficiency of enterprises [5]. Moreover, the Chinese version of SOX system has a significant promoting effect on the improvement of financial performance of listed companies [6]. However, the internal control of some enterprises has not been really implemented. For instance, the auditors issued the negative audit reports of internal control on "Guitang Shares" and other enterprises [7]. Therefore, enterprises should still attach great importance to strengthening the construction of internal control standard system [8].

In the "Application Guidelines for Internal Control No. 10—Research and Development", it is proposed that enterprises should attach importance to the research and development, combine their own development strategy, according to the requirements of market development and technological progress, formulate their scientific plans of research and development. The economic growth theory believes that the technological progress is an important driving force for the economic growth, and the key to the technological progress lies in the scientific and technological innovation. Enterprises are the main body of scientific and technological innovation. Enterprises are the main body of scientific and technological innovation. Enterprises are the output of innovation is the key to improving the driving force of economic growth [9]. As the output of innovation investment is characterized by uncertainty and untimeliness [10, 11], the sound internal control is needed to prevent those possible unclear R&D positioning, lack of capital management, and other moral hazard and adverse selection. In the current stage of Chinese economic development, enterprises need to strengthen the construction of internal control, to promote the steady development of capital market and national economy. Then, under the current background of internal control construction, how does internal control affect the status of technological innovation?

Meanwhile, the non-exclusivity of knowledge should be reasonably utilized to balance the private benefits and social benefits of innovators in the process of continuous operation [12]. The "Application Guidelines for Internal Control No. 4—Social Responsibility" has embedded CSR into the system of internal control. This guideline clearly states that enterprises should fulfill their social obligations and responsibilities in the process of operation and development, including production safety, product quality, environmental protection, resource conservation, promotion of employment, and protection of employees' rights and interests. CSR refers to the responsibilities of enterprises to the creditors, government, customers, employees, community and other stakeholders as well as the environment according to a set of institutional arrangements while assuming economic responsibilities to shareholders [13]. Then, at the present stage, while promoting enterprises to realize the long-term development strategy, what impact does the construction of internal control have on CSR performance? Moreover, from the stakeholder theory, enterprises should actively pay attention to the expectations and



appeals of different stakeholders. CSR activities play an important role in the sustainable development [14]. So, in the context of Chinese economic development entering a new normal, do CSR activities significantly promote technological innovation to further improve their own sustainable development?

Based on the realistic consideration of the relationships between internal control and technological innovation, internal control and CSR performance, and CSR activities and technological innovation, from the perspective of the effectiveness of internal control, this study analyzes the influence of internal control on enterprise innovation performance and CSR performance. On this basis, this study explores the possible intermediary effect of CSR activities in the process of internal control influencing innovation performance, so as to strengthen the construction of internal control and CSR practices, and provide some empirical evidence and valuable suggestions for improving innovation performance under the background of current economic transformation. Fig 1 presents the research train of thought. Where, the symbol "①" represents the effect of internal control on innovation performance. "②" refers to the influence of internal control on CSR. And "(3)" indicates the possible intermediary effect of CSR activities between internal control and innovation performance. The remainder of this paper is organized as follows. Section 2 presents the literature review, theoretical basis, and research hypothesis. Section 3 presents the data source, variable definition, and model setting. Section 4 discusses the descriptive statistics and correlation of variables. Section 5 presents the model regression analysis. Section 6 develops the further analysis. Section 7 develops the robustness test. And section 8 concludes the study.

The possible contribution of this study is as follows. In the past, scholars tend to study the relationship between internal control and innovation performance. Some scholars have shown that internal control effectively prevents various risks in the process of technological innovation through the reasonable risk assessment, control and prevention [15, 16]. Besides, the existing literature tends to study the relationship between internal control and CSR performance. For instance, Ntim and Soobaroyen (2013) [17], Li et al. (2019) [18] believe that the partial content of internal control, as a specific system of corporate governance, still has a positive effect on CSR performance. However, there are few studies on the integration of internal control, CSR activities and innovation performance. On basis of the existing literature on the relationship between corporate governance and innovation performance [19, 20], this study introduces an intermediary variable–CSR performance, and explores deeply the intermediary



Fig 1. The research train of thought.

https://doi.org/10.1371/journal.pone.0234506.g001



effect of CSR activities between internal control and innovation performance. This study expands the mechanism of internal control, CSR activities and innovation performance, and enriches the existing literature on the relationship between internal control and innovation performance.

The possible practical significance of this study is mainly focused on the following aspects. (1) Internal control is an important factor influencing the innovation performance and CSR activities. To strengthen the innovation performance and CSR activities, enterprises should consider enhancing the effectiveness of internal control. (2) The effect of internal control on innovation performance is subject to different property rights. In non-state-owned enterprises, internal control plays a positive role in promoting innovation performance. However, in state-owned enterprises, internal control has not yet shown a significant promoting effect on innovation performance. (3) CSR activities take an intermediary effect between internal control and innovation performance, but this intermediary effect is only reflected in non-state-owned enterprises.

2. Literature review, theoretical basis, and research hypothesis

2.1 Internal control and innovation performance

The essence of innovation is the innovation of human knowledge, science and technology. However, under the condition of information asymmetry, the insiders' opportunistic behaviors lead to the deviation of innovation investment from maximizing shareholders' interests, leading to the low innovation efficiency of enterprises [21]. The promotion of enterprises' innovation ability depends on the effective input mechanism, and the integration of human capital and material capital into the innovation activities is often determined by the internal system of enterprises [19]. The influence of internal control on innovation performance can be divided into two viewpoints, namely the promotion theory of internal control and the paradox of internal control. The promotion theory of internal control holds that the strengthening of internal control contributes to the improvement of technical innovation output [22, 23]. The paradox of internal control holds that the excessive institutionalization often leads to the rigid management, which is not conducive to the implementation of flexible technological innovation [24, 25]. According to Liu et al. (2018) [26], the internal risk management and control ability is significantly and negatively correlated with innovation efficiency, indicating that the paradox of internal control may exist in innovation-oriented enterprises. And on the whole, internal control has a promoting effect on innovation input and innovation performance, but the promoting force is weak [27]. But internal control is implemented late in China, and the overly strict system is not established. The promotion theory of internal control may be more suitable for Chinese national conditions [28]. Also, Chen et al. (2018) [20], Yang et al. (2019) [29], Li and Shi (2019) [30] affirm the appropriateness of the promotion theory of internal control.

The behavior control has a positive moderating effect on the relationship between experimental learning and indigenous innovation [31]. Chinese "Application Guidelines for Internal Control No. 10—Research and Development" aims to promote enterprises to achieve the independent innovation, enhance their core competitiveness, effectively control the risk of R&D activities, and promote the strategy of long-term development. Internal control can effectively reduce the uncertainty of technological innovation by evaluating, controlling and preventing risks [15]. Shi et al. (2014) [32] believe that on the basis of improving the operation mechanism of internal control, enterprises should take the continuous innovation as the long-term development objective, stimulate employees' sense of innovation, and then actively identify, digest, absorb and integrate the cutting-edge knowledge. Enterprises should attach importance to



R&D activities, scientifically formulate R&D plans according to the requirements of market development and technological progress, and strengthen the management of the whole process of R&D activities. On basis of standardizing R&D behaviors, the transformation and effective utilization of R&D results should be promoted to continuously improve the independent innovation abilities of enterprises. Internal control is an internal management system, path and intermediate mechanism to achieve the governance objectives, which can effectively alleviate the agency problem and information asymmetry at the operation level of R&D investment [21, 33]. In the "Application Guidelines for Internal Control No. 10-Research and Development", the approval of R&D project, R&D personnel allocation, management of R&D process, and transformation of R&D results are specified in detail. Through the strict monitoring and intervention, budget and assessment mechanism, the unreasonable R&D projects in the innovation activities can be restrained, and the strategic positioning and risk tolerance boundary of an organization can be clearly conveyed to the employees. Furthermore, it builds a good internal environment for innovation activities and stimulates innovation at all levels within the organization. Therefore, as an important influencing factor of technological innovation, the effective internal control shall play a positive role in promoting innovation performance.

Based on the above analysis, the following research hypothesis is proposed.

Hypothesis 1. The improvement of the effectiveness of internal control will significantly improve the innovation performance of enterprises.

2.2 Internal control and CSR performance

Enterprises should combine their own interests with employees' health and social progress while pursuing entrepreneurs' profits. The goal of corporate governance has experienced a process from the maximization of profit to the maximization of shareholder equity, and then to the maximization of stakeholder benefit. The stakeholder theory emphasizes that while promoting their own profits, enterprises should take into account the corresponding responsibilities for employees, consumers, communities and social development, which achieves a reasonable integration of corporate interests and social effects [34, 35]. CSR is a strategic move that helps to change stakeholders' perceptions of enterprises and expectations of performance, deliver signals of performance improvement, and divert society's attention to enterprises [36]. Barnett and Salomon (2012) [37] found that the relationship between CSR performance and enterprises' financial performance is U-shaped, enterprises with low CSR performance have higher financial performance than enterprises with moderate CSR performance, but enterprises with high CSR performance have the highest financial performance. CSR input can convert to higher benefits only when it generates a solid relationship between enterprises and their stakeholders [38]. When the power supervision between the governance and the management is put in place to achieve higher internal governance, it is conducive to improving CSR performance. Internal control promotes the optimization of capital allocation efficiency, thus promoting the realization of sustainable growth. However, Qin (2019) [39] believes that there is no significant relationship between internal control and CSR activities. But the risk of social responsibility arising from stakeholders must not be ignored. The uncontrolled risk of social responsibility will have a serious adverse impact on the sustainable growth of enterprises [40]. The risk derived from CSR aggravates the risk level of enterprises and broadens the risk boundary faced by enterprises.

Hao et al. (2018) [1] believe that internal control has a significant and partial moderating effect between CSR and stock price crash risk. For enterprises, it is necessary to effectively control the significant behavioral risks that deviate from their established objectives [41]. The significance of CSR activities lies in the transcendence of the governance model of shareholder



supremacy. Enterprises should build a governance mechanism to actively carry out CSR activities on basis of reshaping the governance structure to strengthen their internal system. The effective internal control not only alleviates the conflict of interest caused by those uncoordinated governance structure, but also reasonably guarantees the efficiency of business activities, the reliability of financial reports and the compliance with laws and regulations. The "Application Guidelines for Internal Control No. 4-Social Responsibility" clearly states that enterprises should strengthen the construction of internal control in production safety, product quality, environmental protection, resource conservation, employment promotion and employee interests. This implies that CSR performance has been integrated into the construction process of internal control. Thereby, the effective internal control can help make up for the lack of CSR activities, and exert a positive effect on improving the disclosure quality of CSR information [42, 43]. The effective internal control alleviates CSR risks, protects the legitimate rights and interests of stakeholders, and promotes the successful realization of the strategic objectives of CSR practices. Generally, the internal control can be taken as an institutional system of risk management. When the operation of internal control is effective, it can prevent the improper behaviors that damage the reputation and image, and avoid the negative events that damage CSR practices, to improve the realistic performance of CSR activities. Internal control can promote enterprises to fulfill their social responsibilities by reducing agency costs [18]. The objectives of internal control are in line with the expectations and demands of stakeholders. Therefore, the effective internal control can enhance the practical value of CSR activities. Furthermore, on basis of ensuring the effective implementation of internal control, enterprises will actively carry out CSR activities and improve their CSR performance.

Based on the above analysis, the following research hypothesis is proposed.

Hypothesis 2. The enhancement of the effectiveness of internal control is conducive to the significant improvement of CSR performance.

2.3 Internal control, CSR and innovation performance

As an endogenous monitoring mechanism, the effective internal control alleviates the problem of information asymmetry between enterprises and stakeholders by fully linking enterprises and markets [44-46]. Also, the effective internal control strengthens the efficiency and quality of information communication, forms a good system of division of responsibilities and checks and balances of power, alleviates the conflict between enterprises and stakeholders, and improves the protection of the rights and interests of stakeholders, thus promoting enterprises to actively improve CSR performance [17, 18]. CSR activities mean to satisfy the value demands of stakeholders, thus maintaining the relationship between enterprises and stakeholders. The social responsibility behavior based on stakeholder orientation is conducive to establishing and maintaining a good relationship between enterprises and stakeholders, and supports enterprises to obtain the necessary operational resources from stakeholders [47, 48]. Meanwhile, the production of products with CSR attribute enables enterprises to realize a differentiation strategy, and CSR activities become a resource with competitive advantages [49]. The improvement of CSR performance helps to obtain the information and resources needed in the process of technological innovation [50, 51]. Wang et al. (2019) [52] believe that there is a positive association between CSR and innovation performance in high-polluting firms. Under the circumstance that enterprises choose CSR as a competitive tool, CSR has a significant and positive effect on innovation performance [53]. However, Li et al. (2018) [51] believe that CSR activities have an inverted U-shaped influence on enterprise technological innovation, which can effectively promote enterprise innovation within a certain critical point, but hinder enterprise innovation beyond that critical point.



The activities of technological innovation run through the whole process of R&D decision-making, R&D input, product testing and final innovative products [54]. In the process of technological innovation, the design, testing and use of innovation links are inseparable from the support of stakeholders. Enterprises can obtain more diversified sources of innovative knowledge by assuming CSR to diversified stakeholders [55]. Moreover, through CSR practices, enterprises establish a direct relationship with external stakeholders such as the government, suppliers and customers, thus expanding the sources of external information and knowledge [56]. Enterprises' acquisition and utilization of knowledge promote the development of technological innovation ability. The explicit or implicit knowledge acquired by enterprises from stakeholders plays a key role in the construction of innovation ability [57]. For instance, customers take a more direct role in the reverse catalysis of technological innovation, which can provide the source power for enterprise innovation and play a key role in the early market stage of new products [58]. The external stakeholders often have some new, non-redundant professional knowledge and skills, which can be used to supplement the internal knowledge of enterprises, thus helping to improve the technological innovation of enterprises [59, 60]. Therefore, CSR activities promote a good relationship between enterprises and external stakeholders, and then help enterprises acquire the knowledge and technology owned and accessible by these stakeholders, so as to improve the performance of technological innovation.

Meanwhile, both attitudes and need statements of jobs are affected by informational social influence [61]. Through the active CSR practices, enterprises can convey superior values to the market of technical personnel, which shows that they pay great attention to and attach importance to the interests of stakeholders. In this case, it can make job-seekers feel the expected superior working atmosphere, promote enterprises to introduce more high-quality technical R&D personnel, and reserve human capital for improving the innovation performance. Stakeholders derive functional, psychological and value satisfaction benefits from CSR activities [62]. From the "people-oriented" concept, as the stakeholders of enterprises, R&D personnel are treated fairly and with high quality, and become the direct beneficiaries of CSR activities. In the process of CSR activities, R&D personnel acquire practical skills and experience, and apply relevant skills and experience to their work to obtain the functional benefits. When CSR activities involve the life field concerned by R&D personnel, they connect the work and life through CSR activities to achieve a benign transformation between the work environment and living environment, relieve pressure and discomfort at work, and gain the psychological benefits. Moreover, CSR activities convey shared values to R&D personnel, enabling them to express themselves more freely in their work [63], so as to obtain the value satisfaction benefits. Therefore, enterprises integrate the expectations of R&D personnel into CSR strategy, enhance their recognition and loyalty to the organization, motivate their work enthusiasm and innovation ability, so as to select high-quality innovation projects, formulate scientific innovation plans, and ultimately improve the efficiency of transforming R&D input into innovation results. When enterprises strategically undertake social responsibility, they achieve growth through product and process innovation [2]. Furthermore, considering the hypothesis that the improvement of the effectiveness of internal control significantly strengthens CSR performance (hypothesis 2 above), this study forms an intermediate action path for internal control to affect enterprise innovation performance, that is, CSR activities take an intermediary effect in the process of internal control to improve the innovation performance.

Based on the above analysis, the following research hypothesis is proposed.

Hypothesis 3. CSR activities take an intermediary effect significantly between the effective internal control and enterprise innovation performance.



7/31

3. Data source, variable definition and model setting

3.1 Data source

No doubt that the intensity of R&D investment affects enterprise innovation performance. The disclosed data on the intensity of R&D investment (i.e. the proportion of R&D investment in operating revenue) are more concentrated in the period after 2011. In view of the availability of data, those companies listed on the Shanghai and Shenzhen stock markets are selected as the research sample during 2012–2017. The data are obtained from China stock exchange database, Wind database, and DIB internal control and risk management database. The sample data shall be excluded according to the following criteria. (1) Financial and insurance enterprises. (2) ST, *ST class. (3) Lack of indicator data. On this basis, the continuous variables are Winsorized with two-way 1% quantiles, to avoid the adverse effect of abnormal observations on the analyses.

3.2 Variable definition

1. Interpreted variable. (1) For the evaluation of innovation performance, the number of patent applications can reflect the true innovation ability of enterprises [64]. And the data of patent application is reliable [65]. With reference to the research of Guan and Gao (2009) [66], Berchicci (2013) [67], Maggitti et al. (2013) [68], Tian and Wang (2014) [69], the number of patent applications of enterprises is adopted to measure the innovation performance. That is the sum of the number of inventions, utility models and appearance designs in a year. And the variable symbol is denoted as "PATENT". The more the number of patent applications annually, the higher the performance of innovation output.

(2) For the measure of CSR performance, Shanghai stock exchange in China issued the "Notice on Strengthening the Social Responsibility of Listed Enterprises" in 2008, aiming to guide listed enterprises to actively fulfill their social responsibilities, attach importance to the common interests of stakeholders, and contribute to the construction of harmonious society and the sustainable development of social economy. The notice puts forward the indicator of "social contribution value per share" (SCPS) for the first time, providing a new and important benchmark for the comprehensive and objective evaluation of enterprises' value creation. The disclosure of SCPS information helps the public to have a more comprehensive understanding of the true value that enterprises create for their shareholders, employees, customers, creditors, communities and the society as a whole. Therefore, from the authority and universality of indicators, this study refers to the research design of Fan et al. (2014) [70] and Li et al. (2018) [71], and adopts SCPS as the measurement index of CSR performance. In this regard, SCPS is determined as formula (1).

$$SCPS = C/S$$
 (1)

For formula (1), $C = C_1+C_2+C_3+C_4+C_5-C_6+C_7+C_8-C_9$. The meanings of $C_1 \sim C_5$ are the net margin, income tax expense, business tariff and annex, cash payments to and for employees, employee compensation payable at the end of this period, respectively. C_6 represents the employee compensation payable at the end of last term. The meanings of C_7 , C_8 are the finance charges and donation outlay, respectively. And C_9 represents the sewage charges and cleaning charges. In the numerator of formula (1), the data of "donation outlay" are manually collected from the "non-operating expenses" itemized in the notes to the financial statements. The data of "sewage charges and cleaning charges" are collected manually from the detailed item of "administrative expenses" in the notes to the financial statements. The denominator S of formula (1) represents the average of the total number of shares at the beginning and end of this period.



2. Explanatory variable. Driven by control objectives, the effectiveness of internal control includes the effectiveness of design and the effectiveness of operation. It includes not only the effectiveness of preventing, finding and correcting the major misstatements in the financial reports, but also the effectiveness of restraining the executive behaviors and thereby reducing agency costs [72]. Since the release of the DIB internal control index in 2011, it has been widely recognized by the academia, practice and regulatory departments. Based on the status quo of internal control implemented by listed enterprises, the basic index of internal control is designed in accordance with the realization degree of the five control objectives of compliance, report, asset safety, operation and strategy. Furthermore, internal control index is formed that comprehensively reflects the level of internal control and ability of risk control of listed enterprises. Therefore, for the explanatory variable in the models—the effectiveness of internal control (IC), the internal control index of DIB listed enterprises. The greater is the indicator to evaluate the effectiveness of internal control of enterprises. The greater is the index, the more effective is the implementation of internal control.

3. Control variable. In terms of the selection of control variables, given that R&D investment intensity reflects the amount of resources invested in innovation business, and at home and abroad, there are three main ways to measure R&D investment intensity, which are the proportions of R&D investment in operating revenue, market value and total assets. In this study, the percentage of R&D investment in operating revenue is adopted to evaluate the R&D investment intensity, to investigate its influence on enterprise innovation performance. The reason for this adoption is that enterprises are more willing to carry out technological innovation only on basis of the income, and this approach is more in line with the principle of matching income and expenditure. Meanwhile, the degree of equity concentration will affect the principal-agent relationship among stakeholders and determine the distribution of control rights of listed enterprises. Therefore, the ownership concentration (SHRCR) is taken as a control variable to evaluate its possible impact on enterprise innovation performance and CSR practices. Besides, the higher growth rate of operating revenue implies the strong competitiveness of products to some extent. Through the profits obtained, enterprises can carry out the R&D activities of new products to continuously meet the needs of consumers in the market, which is conducive to the realization of sustainable development and has an impact on CSR activities. Thus, the influence of sales growth rate (GROWTH) is controlled. Meanwhile, the relevant study of Chen et al. (2018) [20] is referenced, the influences of return on total assets (ROA), equity ratio (LER), size of the board of supervisors (SUPERVISOR), double duty (DUAL), executive compensation (LNSALARY), comprehensive leverage (DTL), asset size (LnASSET), audit opinion (AUDIT) and property attribute (STATE) are controlled. The industry effect and annual effect are also controlled in the regression analyses. Table 1 presents the name and calculation method of each variable.

3.3 Model setting-up

In order to verify the rationality of the hypotheses mentioned above, the relevant studies of Hazarika et al. (2012) [72], Chen et al. (2018) [20], Li et al. (2018) [71] are referenced, the following models 1, 2 and 3 are constructed. The regression estimation of panel data for parameters is conducted, respectively, to test the above hypotheses 1, 2 and 3. Furthermore, in order to avoid the adverse impact of dimensional differences between the explained variable and explanatory variable on the results, in model 1—model 3, the value of the explanatory variable (IC) is taken as 1, when the DIB index of an enterprise in this period is higher than that of last period, indicating that the effectiveness of the enterprise's internal control tends to improve.



Nature	Symbol	Name	Calculation method
Explained	PATENT	Innovation performance	The sum of the applications of inventions, utility models and appearance designs in a year
variable	SCPS	CSR performance	Social contribution per share. The calculation method is shown in formula (1) .
Explanatory variable	IC	Effectiveness of internal control	DIB · internal control index of listed companies
Control variable	R&D	Intensity of R&D investment	The proportion of R&D investment in operating revenue
	ROA	Return on total assets	Net profit/average total assets
	LER	Equity ratio	Liabilities/owners' equity
	GROWTH	Sales growth rate	(Current operating income-last operating income)/last operating income
	SHRCR	Ownership concentration	Shareholding ratio of the top 10 shareholders
	SUPERVISOR	Size of the board of supervisors	Total number of supervisors
	DUAL	Double duty	When the chairman concurrently serves as the general manager, the value is 1; otherwise 0
	LNSALARY	Executive compensation	Natural logarithm of total compensation of directors, supervisors and top three executives
	DTL	Comprehensive leverage	Financial leverage × operating leverage
	LnASSET	Asset size	Natural logarithm of total assets at the beginning of the year
	AUDIT	Audit opinion	Dummy variable set according to the type of audit opinion in the disclosed annual audit report. Where, the value of standard unreserved opinion is 1; otherwise 0
	STATE	Property attribute	Dummy variable, if state-owned enterprise, 1; otherwise 0
	YEAR	Year	The annual effect
	IND	Industry	Industry effect. According to the "Guidelines on Industry Classification of Listed Companies (2012 revision)" issued by China Securities Regulatory Commission, the sample enterprises are divided into 17 industries, and a total of 16 industry dummy variables are set
	ε		Random disturbance term

Table 1. Variable name and definition.

https://doi.org/10.1371/journal.pone.0234506.t001

Otherwise, the value is 0, indicating that the effectiveness of internal control remains unchanged or decreases. Meanwhile, in model 1—model 3, the value of the explained variable (LnPATENT) is the natural logarithm of 1 plus the total number of annual patent applications. Besides, to alleviate the endogenous problem caused by the reverse causality, the control variables including R&D, ROA, LER, GROWTH, LNSALARY, DTL and AUDIT are taken as the first-order lag values.

Model 1.

$$LnPATENT_{i,t} = \alpha_0 + \alpha_1 IC_{i,t} + \alpha_2 R\&D_{i,t-1} + \alpha_3 ROA_{i,t-1} + \alpha_4 LER_{i,t-1} + \alpha_5 GROWTH_{i,t-1} + \alpha_6 SHRCR_{i,t} + \alpha_7 SUPERVISOR_{i,t} + \alpha_8 DUAL_{i,t} + \alpha_9 LNSALARY_{i,t-1} + \alpha_{10} DTL_{i,t-1} + \alpha_{11} LnASSET_{i,t} + \alpha_{12} AUDIT_{i,t-1} + \alpha_{13} STATE_{i,t} + \alpha_{14} \sum_{t} YEAR + \alpha_{15} \sum_{t} IND + \varepsilon_{i,t}$$
(2)

Model 2.

$$SCPS_{i,t} = \beta_0 + \beta_1 IC_{i,t} + \beta_2 R \&D_{i,t-1} + \beta_3 ROA_{i,t-1} + \beta_4 LER_{i,t-1} + \beta_5 GROWTH_{i,t-1} + \beta_6 SHRCR_{i,t} + \beta_7 SUPERVISOR_{i,t} + \beta_8 DUAL_{i,t} + \beta_9 LNSALARY_{i,t-1} + \beta_{10} DTL_{i,t-1} + \beta_{11} LnASSET_{i,t} + \beta_{12} AUDIT_{i,t-1} + \beta_{13} STATE_{i,t} + \beta_{14} \sum_{t} YEAR + \beta_{15} \sum_{t} IND + \varepsilon_{i,t}$$

$$(3)$$



Model 3.

$$\begin{aligned} \text{LnPATENT}_{i,t} &= \gamma_0 + \gamma_1 \text{IC}_{i,t} + \gamma_2 \text{SCPS}_{i,t} + \gamma_3 \text{R} \& \text{D}_{i,t-1} + \gamma_4 \text{ROA}_{i,t-1} + \gamma_5 \text{LER}_{i,t-1} + \gamma_6 \text{GROWTH}_{i,t-1} + \\ & \gamma_7 \text{SHRCR}_{i,t} + \gamma_8 \text{SUPERVISOR}_{i,t} + \gamma_9 \text{DUAL}_{i,t} + \gamma_{10} \text{LNSALARY}_{i,t-1} + \gamma_{11} \text{DTL}_{i,t-1} + \\ & \gamma_{12} \text{LnASSET}_{i,t} + \gamma_{13} \text{AUDIT}_{i,t-1} + \gamma_{14} \text{STATE}_{i,t} + \gamma_{15} \sum_{t} \text{YEAR} + \gamma_{16} \sum_{t} \text{IND} + \boldsymbol{\varepsilon}_{i,t} \end{aligned}$$
(4)

Model 1—model 3 are also used to examine the possible intermediary effect of CSR activities in the process of internal control promoting innovation performance, which can be described by the diagram of intermediary effect (Fig 2). Where, the coefficient α_1 in model 1 is the total effect of internal control (IC) on innovation performance (LnPATENT). The coefficient β_1 in model 2 refers to the effect of internal control (IC) on the intermediary variable— Social contribution per share (SCPS). The coefficient γ_2 in model 3 refers to the effect of the intermediary variable (SCPS) on the explained variable—innovation performance (LnPA-TENT) after the effect of internal control (IC) is controlled. And in model 3, the coefficient γ_1 is the direct effect of internal control (IC) on the explained variable (LnPATENT) after the effect of the intermediary variable (SCPS) is controlled.

4. Descriptive statistics and correlation of variables

4.1 Descriptive statistics

Table 2 reports the descriptive statistical results of variables. For the explained variable in model 1 and model 3, the standard deviation of PATENT is 167.710, and the minimum



Fig 2. The diagram of intermediary effect.

https://doi.org/10.1371/journal.pone.0234506.g002

Variable	Mean	Median	Maximum	Minimum	Deviation	Observations
PATENT	88.154	35.000	1138.000	1.000	167.710	4308
LnPATENT	3.533	3.555	7.037	0.000	1.381	4308
SCPS	1.379	1.054	6.230	-0.209	1.154	4304
IC	649.341	673.015	863.140	0.000	128.960	4308
R&D	4.424	3.515	26.900	0.070	4.259	4308
ROA	0.044	0.036	0.214	-0.116	0.051	4308
LER	1.080	0.767	6.330	0.066	1.059	4308
GROWTH	0.154	0.107	2.042	-0.376	0.326	4308
SHRCR	55.208	55.132	87.412	23.472	14.306	4308
SUPVERSIOR	3.705	3.000	7.000	3.000	1.124	4308
DUAL	0.244	0.000	1.000	0.000	0.429	4308
LNSALARY	14.451	14.411	16.193	13.009	0.639	4308
DTL	2.514	1.531	23.498	-1.984	3.401	4308
LnASSET	22.247	22.050	26.054	20.223	1.198	4308
AUDIT	0.984	1.000	1.000	0.000	0.126	4308
STATE	0.426	0.000	1.000	0.000	0.495	4308

Table 2. The descriptive statistics table of variables.

https://doi.org/10.1371/journal.pone.0234506.t002

(maximum) of PATENT is 1 (1138), indicating that the number of annual patent applications of different sample enterprises is quite different. After the number of patent applications is taken into the natural logarithm, the standard deviation of LnPATENT is 1.381, and the mean (median) of LnPATENT is 3.533 (3.555). The variation amplitude of this variable tends to decrease, which can avoid the adverse effect on the following regression analyses. For the explained variable in model 2, the standard deviation of SCPS is 1.154, and the mean (median) of SCPS is 1.379 (1.054), indicating that the sample enterprises have assumed their due social responsibilities to different degrees, and CSR performance of the sample enterprises has a great variability. However, the minimum of SCPS is -0.209, implying that the existence of negative CSR performance in sample enterprises, and there may be some negative response to CSR activities, which will have a negative impact on the overall reputation of enterprises.

For the explanatory variable in model 1—model 3, the mean (median) of IC is 649.341 (673.015), indicating that after the promulgation and implementation of the "Supporting Guidelines for Enterprise Internal Control", the implementation of internal control and the standardization and integrity of information disclosure are generally good, and most enterprises have realized the importance of effective operation of internal control. However, the standard deviation of IC is 128.960, and the maximum (minimum) of IC is 863.140 (0.000). The effectiveness of internal control in the sample enterprises is uneven. During the observation period, there are some cases of internal control failure in some enterprises, and the effectiveness of internal control of these enterprises needs to be further improved.

For the control variables, the mean (standard deviation) of R&D is 4.42% (4.259), and the maximum (minimum) of R&D is 26.90% (0.07%), indicating that the proportion of R&D input in the operating revenue varies to a certain extent, and the emphasis of different sample enterprises on R&D business is obviously different. The minimum of ROA is -0.116, and the standard deviation of LER (GROWTH) is 1.059 (0.326). The above results show that the operating income, liabilities and growth of listed enterprises in China are different. The median of SHRCR is 55.13%, and the average shareholding ratio of top 10 shareholders in the sample enterprises is 55.21%. The mean of DUAL is 24.37%, and the board of supervisors shall be composed of at least 3 supervisors and at most 7 supervisors. The standard deviations of



LNSALARY, DTL and LnASSET are 0.639, 3.401 and 1.198, respectively. In addition, 42.60% of the sample enterprises are state-owned. The mean of AUDIT is 0.984. During the observation period, the external auditors take a positive attitude towards the legality and fairness of financial reports of nearly 98.40% of the sample enterprises, which ensures the reliability of the data used in this study.

4.2 Correlation of variables

Table 3 presents the correlation coefficients of variables. For model 1 and model 3, the correlation coefficient between the explanatory variable (IC) and the explained variable (LnPATENT) is 0.102 (p < 0.01). This result, to some extent, supports hypothesis 1 above, indicating that the effective internal control is conducive to improving enterprises' innovation performance. For model 2, the correlation coefficient between the explanatory variable (IC) and the explained variable (SCPS) is 0.243 (p < 0.01), implying that internal control can be a driving force for CSR practices. Hypothesis 2 above is initially supported. For model 3, the correlation coefficient between SCPS and LnPATENT is 0.277 (p < 0.01), suggesting that the strengthening of CSR practices can be a favorable factor to enhance enterprises' innovation performance. Meanwhile, with the elaboration of hypothesis 2, CSR activities are expected to present an intermediary effect significantly between internal control and innovation performance. Hypothesis 3 above is preliminarily supported.

Among the control variables in model 1 and model 3, ROA (0.062), LER (0.142), GROWTH (0.052), SHRCR (0.094), SUPERVISOR (0.140), LNSALARY (0.284), LnASSET (0.443), AUDIT (0.055), STATE (0.113) are positively and significantly (p < 0.01) correlated with the explained variable (LnPATENT), respectively. DUAL (-0.049) is negatively and significantly (p < 0.01) correlated with the explained variable (LnPATENT). These reported results of correlation coefficients ensure the rationality of model 1 and model 3 above. Also, among the control variables in model 2, ROA (0.461), LER (0.141), GROWTH (0.158), SHRCR (0.242), SUPERVISOR (0.189), LNSALARY (0.349), LNASSET (0.362), AUDIT (0.071), STATE (0.213) are positively and significantly (p < 0.01) correlated with the explained variable (SCPS), respectively. R&D (-0.106, p < 0.01), DUAL (-0.078, p < 0.01), DTL (-0.100, p < 0.05) are negatively and significantly (p < 0.01) correlated with the explained variable (SCPS), respectively. These reported results of correlation coefficients ensure the rationality of model 2 above.

Also, it is shown in Table 3 that the maximum absolute value of the correlation coefficients is 0.437. This maximum exists between LnASSET and LER, which is less than the threshold of 0.800. Taken together, it is shown that there is no severe multicollinearity in the model 1— model 3, which provides a reliable guarantee for the regression analyses below.

5. Model regression analysis

The descriptive statistics for single variable and correlation coefficients between variables are presented above. But they are only preliminary analyses because the other factors that affect the explained variables are not included. The data type used in this study is panel data. Panel data analysis has a certain advantage for alleviating the endogenous problem caused by missing variables that do not vary with time. The specific regression analysis methods include the mixed OLS method, fixed effect model and random effect model. LSDV method is adopted to analyze model 1—model 3, and the null hypothesis that "the coefficients of all individual dummy variables are zero" is rejected, indicating the existence of individual fixed effect. The robust Hausman tests for model 1—model 3 show that the Sargan-Hansen χ^2 are 315.722 (p = 0.000), 415.706 (p = 0.000), 329.610 (p = 0.000), respectively, indicating that the fixed



		0400	Ģ		104	44.1				1111			11100	
	-	SCPS	- <u>-</u>	R&D	KUA	LER	GROWTH	SHRCR	SUPVERSIOR	DUAL	LNSALARY	DIL	LIASSET	AUDIT
	*	1.000												
** 0.106^{*+} 0.014 1.000 \cdots <	*	0.243^{***}	1.000											
** 0.461^{**} 0.341^{**} 0.050^{**} 1.000 ···index <td>*</td> <td>-0.106^{***}</td> <td>-0.014</td> <td>1.000</td> <td></td>	*	-0.106^{***}	-0.014	1.000										
*** 0.141^{***} 0.105^{***} 0.256^{***} 1.000 1.000 1.00 <th< td=""><td>* * *</td><td>0.461^{***}</td><td>0.341^{***}</td><td>0.050^{***}</td><td>1.000</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	* * *	0.461^{***}	0.341^{***}	0.050^{***}	1.000									
*** $0.13**$ $0.013**$ 0.001 0.242^{**} 0.016 1.000 1.000 1.010 </td <td>* * *</td> <td>0.141^{***}</td> <td>-0.105***</td> <td>-0.266***</td> <td>-0.353***</td> <td>1.000</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	* * *	0.141^{***}	-0.105***	-0.266***	-0.353***	1.000								
*** 0.242^{***} 0.117^{***} 0.157^{***} 1.000 <th< td=""><td>* *</td><td>0.158^{***}</td><td>0.113^{***}</td><td>-0.001</td><td>0.242^{***}</td><td>-0.016</td><td>1.000</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	* *	0.158^{***}	0.113^{***}	-0.001	0.242^{***}	-0.016	1.000							
*** 0.189^{***} 0.004 -0.117^{***} -0.025 0.191^{***} -0.054^{***} 0.060^{***} 1.000 $\cdot \cdot \cdot \cdot$ $\cdot \cdot \cdot \cdot$ $\cdot \cdot \cdot \cdot \cdot$ $\cdot \cdot \cdot \cdot \cdot$ $\cdot \cdot \cdot \cdot \cdot \cdot$ $\cdot \cdot \cdot \cdot \cdot \cdot$ $\cdot \cdot \cdot \cdot \cdot \cdot \cdot$ $\cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot$ $\cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot \cdot$ $\cdot \cdot \cdot$ $\cdot \cdot $	* *	0.242^{***}	0.117^{***}	-0.113***	0.155^{***}	-0.005	0.090^{***}	1.000						
9 ⁺⁺⁺ 0.078 ⁺⁺⁺ 0.018 ⁺⁺⁺ 0.0139 ⁺⁺⁺ 0.0139 ⁺⁺⁺ 0.0139 ⁺⁺⁺ 0.0139 ⁺⁺⁺⁺ 0.0149 ⁺⁺⁺⁺ 1.000 1.)***	0.189^{***}	0.004	-0.117***	-0.025	0.191^{***}	-0.054^{***}	0.060^{***}	1.000					
	9***	-0.078***	-0.001	0.118^{***}	0.049^{***}	-0.139***	0.011	-0.045***	-0.149^{***}	1.000				
18 -0.100** -0.030* -0.021** 0.212** -0.033** -0.051*** 0.062*** -0.043*** -0.126*** 1000 ~ ~ 5*** 0.362*** 0.090** -0.052*** 0.212** 0.083*** 0.005*** 0.043*** -0.126*** 1.000 ~	1*** 1	0.349^{***}	0.150^{***}	0.047^{***}	0.310^{***}	-0.036^{**}	0.065***	0.055^{***}	0.056^{***}	-0.008	1.000			
	18	-0.100^{**}	-0.030^{*}	-0.091^{**}	-0.222**	0.212^{**}	-0.083**	-0.051***	0.062^{***}	-0.043^{***}	-0.126***	1.000		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	3***	0.362^{***}	0.090***	-0.282***	-0.052***	0.437^{***}	-0.087***	0.199^{***}	0.320^{***}	-0.170***	0.366***	0.137^{***}	1.000	
5*** 0.213*** -0.019 -0.170*** -0.296*** -0.093*** 0.065*** 0.376*** -0.288*** -0.001 0.101*** 0.357*** 0.002	5***	0.071^{***}	0.389^{***}	-0.024	0.135^{***}	-0.063***	0.048^{***}	0.036^{**}	-0.006	0.016	0.025	-0.015	-0.014	1.000
	***	0.213^{***}	-0.019	-0.170***	-0.150***	0.296^{***}	-0.093***	0.065***	0.376^{***}	-0.288***	-0.001	0.101^{***}	0.357^{***}	0.002

Table 3. The table of correlation coefficients among variables.

PLOS ONE | https://doi.org/10.1371/journal.pone.0234506 June 11, 2020

** Significant at 5%* Significant at 10%.

https://doi.org/10.1371/journal.pone.0234506.t003

effect model should be adopted instead of random effect. Accordingly, the regression estimation results of model 1—model 3 are shown in Table 4.

5.1 Analyses of model 1's regression results

5.1.1 Analyses of model 1's explanatory variable. As shown in column 2 of Table 4, in model 1, the coefficient on the explanatory variable (IC) is positive and significant (0.046, p < 0.05). This result shows that the more effective the implementation of internal control is, the higher the innovation output of the enterprise is. The improvement of the effectiveness of internal control has a significant and positive impact on the innovation performance. Hypothesis 1 above is verified. The implementation of internal control enables the management to effectively identify the external environmental risks, opportunities and internal resources through the risk assessment, select those high-quality innovation projects and improve the efficiency of innovation output. To improve the innovation ability of enterprises, it is necessary to attach importance to the construction of internal system. There are many factors affecting the innovation ability of enterprises, but internal control is undoubtedly an important factor [73]. Innovation performance is closely related to the internal resource allocation system of enterprises. It is an important measure to improve the innovation ability of enterprises to attach importance to the micro-level system construction of enterprises, including internal control. The effective internal control prevents various risks in the process of technological innovation through the reasonable risk assessment, control and prevention [15, 16].

Variable	Model 1	Model 2	Model 3
	Coefficient (S.E.)	Coefficient (S.E.)	Coefficient (S.E.)
Intercept	-4.791** (1.873)	3.325* (1.819)	-5.289*** (1.841)
IC	0.046** (0.023)	0.064*** (0.017)	0.040* (0.023)
SCPS			0.135*** (0.036)
L.R&D	0.023* (0.013)	0.002 (0.008)	0.019 (0.016)
L.ROA	2.103*** (0.526)	1.358*** (0.364)	1.892*** (0.532)
L.LER	-0.004 (0.055)	0.214*** (0.044)	-0.032 (0.055)
L.GROWTH	-0.028 (0.050)	0.201*** (0.039)	-0.060 (0.051)
SHRCR	0.003 (0.003)	0.015*** (0.003)	0.001 (0.003)
SUPERVISOR	-0.055 (0.047)	-0.050 (0.043)	-0.052 (0.047)
DUAL	0.224*** (0.056)	0.016 (0.042)	0.218*** (0.056)
L.LNSALARY	0.044 (0.065)	0.064 (0.048)	0.035 (0.065)
L.DTL	-0.001 (0.004)	-0.011*** (0.003)	-0.0001 (0.004)
LnASSET	0.319*** (0.070)	-0.173** (0.067)	0.346*** (0.068)
L.AUDIT	-0.007 (0.174)	-0.024 (0.165)	-0.013 (0.169)
STATE	-0.135 (0.209)	0.022 (0.128)	-0.140 (0.207)
YEAR/ IND	YES	YES	YES
# of obs.	3590	3586	3586
Within_R ²	0.117	0.159	0.125
F_Value	15.69***	22.34***	16.18***

Table 4. Statistical results of regression coefficients for model 1-model 3.

*** Significant at 1%

** Significant at 5%

* Significant at 10%.

Robust standard errors in brackets are clustered at the enterprise level.

https://doi.org/10.1371/journal.pone.0234506.t004



5.1.2 Analyses of model 1's control variables. In the control variables, the coefficient on L.R&D is positive and significant (0.023, p < 0.10), indicating that the increase of R&D investment is undoubtedly an important factor to improve innovation performance. Enterprises should formulate their scientific R&D expenditure plans according to the established development strategy. The coefficient on L.ROA is positive and significant (2.103, p < 0.01), indicating that the good accounting earnings can provide more financing for innovation investment. The coefficient on DUAL is positive and significant (0.224, p < 0.01). When the chairman of the board is also the general manager, the senior executives take their own performance evaluation into consideration and try to avoid the inefficiency of innovative resource input, to improve the performance evaluation of senior executives by stakeholders. The coefficient on LnASSET is positive and significant (0.319, p < 0.01), implying that those large-scale enterprises often have strong R&D strength, which is conducive to improving their innovation performance. The estimated coefficients on the other control variables are not statistically significant.

5.2 Analyses of model 2's regression results

5.2.1 Analyses of model 2's explanatory variable. As shown in column 3 of Table 4, in model 2, the coefficient on the explanatory variable (IC) is positive and significant (0.064, p < 0.01), indicating that the improvement of the effectiveness of internal control promotes the performance of CSR activities. Hypothesis 2 above is verified. Enterprises carry out CSR activities through various commercial and social practices, bringing sustainable and just benefits to stakeholders, which promotes the welfare of society, and protects the members of organizations. On basis of the comprehensive monitoring, the effective internal control is adopted to supervise and correct the process of production and operation, so as to ensure that the operational risks are evaded sufficiently. Enterprises strengthen the construction of internal control, and protect the legitimate rights and interests of stakeholders, which to a large extent reflect the basic objectives of social governance. The partial content of internal control, as a specific system of corporate governance, still has a positive effect on the performance of social responsibility [17, 18].

5.2.2 Analyses of model 2's control variables. In the control variables, the coefficient on L.ROA is positive and significant (1.358, p < 0.01), suggesting that the better accounting surplus of enterprises provides more material support for CSR activities, which is conducive to promoting enterprises to carry out CSR activities, which conforms to the resource-based hypothesis. The coefficient on L.LER is positive and significant (0.214, p < 0.01), implying that the creditor governance has a positive governance effect on CSR practices. The coefficient on L.GROWTH is positive and significant (0.201, p < 0.01), indicating that the better the development prospect of enterprises, the better the performance of social responsibility. The coefficient on SHRCR is positive and significant (0.015, p < 0.01). When the degree of equity concentration is high, the interests of major shareholders and the enterprise as a whole tend to be consistent. In order to realize long-term interests, major shareholders tend to encourage enterprises to better carry out CSR activities. Besides, the coefficient on DTL is negative and significant (-0.011, p < 0.01), suggesting that for those enterprises with higher operational risks, senior executives should be urged to improve CSR performance on basis of establishing an effective incentive mechanism while realizing entrepreneur profits. And the coefficient on LnASSET is negative and significant (-0.173, p < 0.05). While pursuing profits, large-scale enterprises should fully pay attention to the legitimate rights and interests of stakeholders, so as to gain a better consumer trust and social image. The estimated coefficients on the other control variables are not statistically significant.



5.3 Analyses of model 3's regression results

5.3.1 Analyses of the intermediary effect. In model 3, the coefficient on IC is positive and significant (0.040, p < 0.10), indicating that the improvement of the effectiveness of internal control will help improve the innovation performance of enterprises. The coefficient on SCPS is positive and significant (0.135, p < 0.01), indicating that the strengthening of CSR practices will be a positive driving factor to enhance innovation performance. In addition to economic interests, enterprises should actively respond to the expectations and appeals of different stakeholders, enhance their social reputation and establish a good corporate image. The good social reputation of enterprises will enhance their competitiveness in an invisible way. In order to effectively maintain an enterprise's competitive advantage, stakeholders supervise and motivate the management to identify the external opportunities and internal resources, formulate the scientific innovation plans, combine the market development with technology development, and then effectively improve their innovation performance. Meanwhile, given that the effect of internal control on social contribution per share is also positive and significant $(\beta_1, p < 0.01)$, in general, CSR activities take an intermediary effect between internal control and innovation performance. Hypothesis 3 above is verified. In addition to economic interests, enterprises should feedback to stakeholders' expectations, supervise and motivate the management to make the scientific innovation plans, so as to effectively improve the level of innovation performance. Moreover, the non-parametric percentile Bootstrap method of deviation correction is adopted. After 1000 runs, the confidence interval corrected of $\beta_1 \times \gamma_2$ for 95% confidence is [0.007, 0.028]. Where, $\beta_1 \times \gamma_2$ is the product of the effect of IC on SCPS and that of SCPS on LnPATENT. Moreover, the effect size of the intermediary effect is about 18.78% (i.e. $\beta_1 \times \gamma_2 / \alpha_1 = 0.064 \times 0.135 / 0.046$).

5.3.2 Analyses of model 3's control variables. In the control variables, the coefficients on L.ROA, DUAL and LnASSET are positive and significant (1.892, p < 0.01; 0.218, p < 0.01; 0.346, p < 0.01). This indicates that these variables have convergent analysis conclusions with the corresponding variables of model 1 in Table 4. The good accounting surplus, appraisal expectations for senior executives and scale advantage are the positive factors to enhance the efficiency of enterprise innovation. The estimated coefficients on the other control variables are not statistically significant.

6. Further analysis

According to the composition of listed enterprises in Chinese capital market, state-owned enterprises occupy a dominant position in terms of scale and number [74]. The effect of internal control on innovation input is influenced by the property attributes of enterprises [75]. The operation mode, operation management and supervision of state-owned enterprises and non-state-owned enterprises are not the same, and there are some differences in the efficiency of technological innovation and CSR activities. Therefore, the property attributes are taken as the basis for grouping test. Based on distinguishing the property attributes of different enterprises, this study conducts the further analyses on model 1—model 3.

6.1 Further analyses of model 1's regression results

For model 1, this study distinguishes between the state-owned and non-state-owned samples. LSDV method is used to analyze model 1, and the null hypothesis that "the coefficients of all individual dummy variables are zero" is rejected, indicating the existence of individual fixed effect. The robust Hausman tests for model 1 show that the Sargan-Hansen χ^2 are 231.172 (p = 0.000), 230.268 (p = 0.000), respectively, indicating that the fixed effect model should be



adopted instead of random effect. Accordingly, the statistical results of the fixed-effect models are shown in columns 2 and 3 of Table 5.

6.1.1 Further analyses of model 1's explanatory variable. As shown in columns 2 and 3 of Table 5, in the state-owned sample, the coefficient on the explanatory variable (IC) is not statistically significant (0.008, p > 0.10). And in the non-state sample, the coefficient on the explanatory variable (IC) is positive and significant (0.068, p < 0.05). These above results suggest that the positive effect of internal control on technological innovation is limited by the property attributes of enterprises, and the improvement of the effectiveness of internal control in state-owned enterprises does not exert a significant effect on the performance of technological innovation. The results show that the enhancement of the effectiveness of internal control is conducive to improving innovation performance, which comes from the supervision and motivation of internal control during the implementation of innovation projects, reducing

Variable	Mo	del 1	Мо	del 2	Мо	lel 3
	State-owned	Non-state	State-owned	Non-state	State-owned	Non-state
	Coefficient (S. E.)	Coefficient (S. E.)	Coefficient (S. E.)	Coefficient (S. E.)	Coefficient (S. E.)	Coefficient (S. E.)
Intercept	-6.141* (3.378)	-3.218 (2.094)	3.245 (3.161)	1.450 (2.170)	-6.937** (3.167)	-3.390 (2.087)
IC	0.008 (0.034)	0.068** (0.031)	0.092*** (0.031)	0.048 ^{***} (0.018)	-0.003 (0.034)	0.066** (0.031)
SCPS					0.168*** (0.047)	0.086* (0.044)
L.R&D	0.006 (0.022)	0.031** (0.015)	0.014 (0.012)	-0.003 (0.009)	-0.008 (0.035)	0.032** (0.015)
L.ROA	1.525* (0.866)	2.487*** (0.668)	2.155*** (0.616)	0.876* (0.445)	1.019 (0.856)	2.459*** (0.669)
L.LER	-0.020 (0.055)	-0.021 (0.095)	0.214*** (0.073)	0.190*** (0.055)	-0.058 (0.054)	-0.035 (0.097)
L.GROWTH	-0.044 (0.072)	-0.016 (0.071)	0.165** (0.065)	0.225*** (0.045)	-0.087 (0.075)	-0.035 (0.069)
SHRCR	0.017*** (0.006)	-0.004 (0.003)	0.024*** (0.007)	0.008*** (0.002)	0.013*** (0.005)	-0.005 (0.003)
SUPERVISOR	0.006 (0.057)	-0.070 (0.062)	-0.057 (0.056)	-0.005 (0.046)	0.015 (0.060)	-0.070 (0.062)
DUAL	0.035 (0.080)	0.305*** (0.075)	0.010 (0.092)	0.006 (0.042)	0.034 (0.081)	0.300*** (0.075)
L.LNSALARY	0.092 (0.092)	-0.073 (0.087)	0.077 (0.077)	0.083 (0.062)	0.079 (0.092)	-0.079 (0.086)
L.DTL	-0.003 (0.005)	0.004 (0.007)	-0.014^{***} (0.005)	-0.006* (0.003)	-0.001 (0.005)	0.005 (0.007)
LnASSET	0.292** (0.121)	0.338*** (0.081)	-0.172 (0.118)	-0.105 (0.081)	0.333*** (0.115)	0.350*** (0.081)
L.AUDIT	-0.143 (0.235)	0.102 (0.234)	-0.077 (0.339)	0.029 (0.105)	-0.128 (0.220)	0.086 (0.235)
YEAR/ IND	YES	YES	YES	YES	YES	YES
# of obs.	1524	2066	1521	2065	1521	2065
Within_R ²	0.132	0.125	0.234	0.110	0.151	0.128
F_Value	7.81***	11.04***	15.65***	9.48***	8.77***	10.75***

Table 5. Statistical results of further analysis for models 1-3.

*** Significant at 1%

** Significant at 5%

* Significant at 10%.

Robust standard errors in brackets are clustered at the enterprise level.

https://doi.org/10.1371/journal.pone.0234506.t005



those inefficient R&D projects, and strengthening the efficiency of innovation activities to improve innovation performance. However, the positive effect of internal control on innovation performance is only prominent in non-state-owned enterprises. And in state-owned enterprises, the implementation of internal control has not yet played an obvious promoting effect on innovation performance. The positive impact of internal control on innovation performance is subject to the property attributes, and the impact of internal control on technological innovation is not significant in the state-owned sample, which may be caused by the unclear ownership and management rights, and the lack of incentive and supervision mechanism [76]. After T test with sample data, it is found that the difference between the mean of internal control index of state-owned enterprises and that of non-state-owned enterprises is 4.927, which is not statistically significant. Therefore, the regulators should urge the stateowned enterprises to play the positive role of internal control in innovation activities, so as to promote the healthier and more stable economic development.

6.1.2 Further analyses of model 1's control variables. In the non-state sample, the coefficient on L.R&D is positive and significant (0.031, p < 0.05). The intensification of R&D investment will be an important factor in improving innovation performance, but this positive effect is only reflected in the non-state sample. The possible reason is that the residual control rights and residual claims of state-owned enterprises are difficult to correspond to the uncertain technological innovation choices, while non-state-owned enterprises can better solve this problem, and the technological innovation efficiency of non-state-owned enterprises is higher than that of state-owned enterprises. In the state-owned and non-state samples, the coefficients on L.ROA are positive and significant (1.525, p < 0.10; 2.487, p < 0.01). In general, a good accounting surplus is the material guarantee to improve innovation performance. In the stateowned sample, the coefficient on SHRCR is positive and significant (0.017, p < 0.01). For state-owned enterprises, it is an effective way to improve innovation performance to enhance the supervision of major shareholders on innovation activities under the circumstance that internal control has not vet exerted a significant promoting effect on innovation performance. In the non-state sample, the coefficient on DUAL is positive and significant (0.305, p < 0.01). This result implies that in non-state-owned enterprises, when the chairman is also the general manager, stakeholders have higher expectation on the performance of senior executives, which to some extent promotes senior executives to enhance innovation performance. Besides, in the state-owned and non-state samples, the coefficients on LnASSET are positive and significant (0.292, p < 0.05; 0.338, p < 0.01). Furthermore, it can be found from the correlation of variables (Table 3) that LnASSET is positively and significantly (0.090, p < 0.01) correlated with IC. While paying attention to strengthen the effectiveness of internal control, larger enterprises have effectively improved their innovation performance. Large enterprises often have a strong competitive advantage in the aspects of cost, market share and brand influence.

6.2 Further analyses of model 2's regression results

For model 2, this study distinguishes between the state-owned and non-state-owned samples. LSDV method is used to analyze model 2, and the null hypothesis that "the coefficients of all individual dummy variables are zero" is rejected, indicating the existence of individual fixed effect. The robust Hausman tests for model 2 show that the Sargan-Hansen χ^2 are 156.420 (p = 0.000), 148.089 (p = 0.000), respectively, indicating that the fixed effect model should be used instead of random effect. Accordingly, the statistical results of the fixed-effect model are shown in columns 4 and 5 of Table 5.

6.2.1 Further analyses of model 2's explanatory variable. In the state-owned and non-state samples, the coefficients on the explanatory variable (IC) are positive and significant



(0.092, p < 0.01; 0.048, p < 0.01). The improvement of the effectiveness of internal control promotes enterprises to actively carry out the necessary CSR practices. The effective internal control has a spillover effect, which not only affects the production and operation of enterprises themselves, but also affects all stakeholders. Hypothesis 2 above is verified again. Enterprises should continue to strengthen the internal governance and form an effective system of internal control, which will be an important safeguard for safeguarding the legitimate rights and interests of stakeholders.

6.2.2 Further analyses of model 2's control variables. In the state-owned and non-state samples, the coefficients on L.ROA are positive and significant (2.155, p < 0.01; 0.876, p < 0.10). Whether in state-owned or non-state-owned enterprises, the good return on assets is the material premise to engage in CSR activities. The coefficients on L.LER are positive and significant (0.214, p < 0.01; 0.190, p < 0.01). Creditor governance has a positive effect on CSR practices. And the coefficients on L.GROWTH are positive and significant (0.165, p < 0.05; 0.225, p < 0.01). The optimistic development prospect of enterprises is often accompanied by the better CSR practices. Also, the coefficients on SHRCR are positive and significant (0.024, p < 0.01; 0.008, p < 0.01). When the enterprise's equity is relatively concentrated, the interests of major shareholders and the overall interests of the enterprise are convergent, which improves CSR performance. Besides, the coefficients on L.DTL are negative and significant (-0.014, p < 0.01; -0.006, p < 0.10). The excessive operational risk will take a negative impact on CSR activities. Therefore, it is a strategic move for the management to control the operation risk at a reasonable level.

6.3 Further analyses of model 3's regression results

For model 3, this study distinguishes between the state-owned and non-state-owned samples. LSDV method is used to analyze model 3, and the null hypothesis that "the coefficients of all individual dummy variables are zero" is rejected, indicating the existence of individual fixed effect. The robust Hausman tests for model 3 show that the Sargan-Hansen χ^2 are 204.646 (p = 0.000), 212.853 (p = 0.000), respectively, indicating that the fixed effect model should be adopted instead of random effect. Accordingly, the statistical results of the fixed-effect model are shown in columns 6 and 7 of Table 5.

6.3.1 Further analyses of the intermediary effect. In the state-owned sample, the coefficient on IC is not statistically significant (-0.003, p > 0.10). And in the non-state sample, the coefficient on IC is positive and significant (0.066, p < 0.05). Once again, the results show that the positive effect of internal control on technological innovation is limited by the property attributes. The internal control of non-state-owned enterprises takes a significant and positive effect on the performance of technological innovation. However, in state-owned enterprises, internal control does not give play to a significant impact on the performance of technological innovation. For state-owned enterprises, the positive effect of internal control on innovation performance should remain the focus of regulators' work in the future.

In the state-owned sample, the coefficient on SCPS is positive and significant (0.168, p < 0.01). In the non-state sample, the coefficient on SCPS is positive and significant (0.086, p < 0.10). The enhancement of CSR performance will be an effective driving factor to improve innovation performance. Meanwhile, for model 1, in the state-owned sample, the coefficient on the explanatory variable (IC) is not statistically significant (0.008, p > 0.10). And in the non-state sample, the coefficient on IC is positive and significant (0.068, p < 0.05). Thus, hypothesis 3 above is verified again. However, the intermediary effect of CSR between internal control and innovation performance is only reflected in the non-state-owned sample. And then, the non-parametric percentile Bootstrap method of deviation correction is adopted.



After 1000 runs, in the non-state sample, the confidence interval corrected of $\beta_1 \times \gamma_2$ for 95% confidence is [0.013, 0.048]. Where, $\beta_1 \times \gamma_2$ represents the product of the effect of IC on SCPS and that of SCPS on LnPATENT. Also, the effect size of the intermediary effect is about 6.07% (i.e. $\beta_1 \times \gamma_2/\alpha_1 = 0.048 \times 0.086/0.068$).

6.3.2 Further analyses of model 3's control variables. For the control variables, in the state-owned sample, the coefficients on SHRCR and LnASSET are positive and significant (0.013, p < 0.01; 0.333, p < 0.01), implying that the moderate equity concentration and scale advantage are favorable motivations to enhance the efficiency of innovation output in state-owned enterprises. And in the non-state sample, the coefficients on L.R&D, L.ROA, DUAL and LnASSET are positive and significant (0.032, p < 0.05; 2.459, p < 0.01; 0.300, p < 0.01; 0.350, p < 0.01), indicating that the abundant R&D resources input, good accounting surplus, performance expectation for senior executives and scale advantage are all positive drivers to strengthen innovation performance in non-state-owned enterprises. This indicates that these above variables have convergent analysis conclusions with the corresponding variables of model 3 in Table 5.

7 Robustness tests

The effectiveness of internal control is influenced by the environment and characteristics of the enterprise. And the characteristic variables of enterprises with high effectiveness of internal control and enterprises with low effectiveness are inevitably quite different. If these characteristic variables also affect enterprise innovation performance and CSR performance, the relationship between internal control and innovation performance observed in the previous whole-sample analyses may be interfered with. Therefore, Abadie and Imbens (2006) [77] Matching Estimator is adopted to construct model 4 based on the multiple dimensions of corporate characteristics and governance status, and the Logistic regression is adopted to estimate the propensity score of the effectiveness of internal control (IC). The dummy variable (DIC) of IC is set as the explained variable of Logistic regression. When the actual value of IC is greater than the industry-annual median, the value of DIC is 1; otherwise, it is 0. And then, in addition to the original control variables as covariates, this study refers to the study of Wang et al. (2017) [78], adds the marketization index of the registration place of listed enterprises (Marketization) as a covariate, and screen out the treatment group and control group. Different from the matching of a single indicator, the propensity score matching (PSM) condenses multiple features into a "propensity score value", thus facilitating the overall matching of multiple features to obtain the same or similar matching samples in terms of the main characteristic variables, so as to reduce the adverse impact of other possible interfering factors on the conclusions. In this study, in accordance with model 4, the Logistic regression is conducted and the propensity scores are calculated. Based on the principle of one-to-one correspondence and near-neighbor matching without replacement, a total of 1536 pairs of paired observations are obtained. And then, the regression analyses of the models are carried out again according to the paired sample.

Model 4.

$$\begin{aligned} \text{DIC}_{i,t} &= \theta_0 + \theta_1 \text{Marketization}_{i,t} + \theta_2 \text{R} \otimes \text{D}_{i,t-1} + \theta_3 \text{ROA}_{i,t-1} + \theta_4 \text{LER}_{i,t-1} + \theta_5 \text{GROWTH}_{i,t-1} + \theta_6 \text{SHRCR}_{i,t} \\ &+ \theta_7 \text{SUPERVISOR}_{i,t} + \theta_8 \text{DUAL}_{i,t} + \theta_9 \text{LNSALARY}_{i,t-1} + \theta_{10} \text{DTL}_{i,t-1} + \theta_{11} \text{LnASSET}_{i,t} + \\ &\theta_{12} \text{AUDIT}_{i,t-1} + \theta_{13} \text{STATE}_{i,t} + \theta_{14} \sum_{t} \text{YEAR} + \theta_{15} \sum_{t} \text{IND} + \boldsymbol{\varepsilon}_{i,t} \end{aligned}$$
(5)



7.1 Robustness tests before properties are distinguished

In view of the differences in the difficulty of technological innovation of enterprises in different industries, it is necessary to consider the influence of industry factors when evaluating innovation performance. Therefore, this study refers to the research design of Li et al. (2018) [51], measures the innovation performance of an enterprise in the current year by dividing the annual number of patent applications by the average of the current year's patent applications in the industry to which the enterprise belongs, which helps to increase the comparability of innovation output among enterprises in different industries. The specific calculation method is shown in formula (6). Where, M_PATENT represents an enterprise's annual innovation performance. In the numerator, P represents the number of patent applications of the industry to which the enterprise belongs in the current period, that is, the quotient between the total number of patent applications of a certain industry and the number of individual enterprises in that industry. Furthermore, the calculated results are taken as the explained variable of model 1 and model 3 for the regression analyses again.

$$M_PATENT = {}^{P}/_{M_P}$$
(6)

Meanwhile, this study takes into account the real social responsibilities of enterprises to stakeholders, which is represented by the cash actually paid to stakeholders under the given income status. Therefore, the research design of Li et al. (2018) [71] is referenced. In this study, the fulfillment of social responsibility (FCSR) is calculated. The specific value of FCSR is the ratio between the cash flow paid for shareholders, creditors, employees, customers, consumers, suppliers, communities, other stakeholders and the average of total shares in the current period. The specific calculation method of FCSR is shown in formula (7). Where, in the numerator of formula (7), M_1 represents the cash paid to distribute dividends or profits. M_2 represents the operating cash charges. M_3 represents the cash paid for goods purchased and services received. And M_6 represents the cash actually paid for taxes and dues. The denominator S is the average of the total number of shares at the beginning and end of the period. In this way, CSR performance is measured again. The larger the calculated value of FCSR, the higher the performance of CSR activities. Furthermore, FCSR is taken as the explained variable of model 2 and the intermediary variable of model 3 for the regression analyses again.

$$FCSR = \sum_{i=1}^{6} M_i / S \tag{7}$$

In accordance with the PSM samples, LSDV method is used to analyze model 1—model 3, and the null hypothesis that "the coefficients of all individual dummy variables are zero" is rejected, indicating the existence of individual fixed effect. The robust Hausman tests for model 1- model 3 show that the Sargan-Hansen χ^2 are 107.432 (p = 0.000), 101.228 (p = 0.000) and 112.548 (p = 0.000), respectively, indicating that the fixed effect model should be adopted. Table 6 reports the robustness test results of the whole sample for model 1- model 3.

As shown in Table 6, in model 1, the coefficient on IC is positive and significant (0.025, p < 0.10), showing that the effective internal control is helpful to improve the efficiency of innovation output. Hypothesis 1 above is verified again. In model 2, the coefficient on IC is positive and significant (0.151, p < 0.01), implying that the effective internal control is conducive to significantly strengthening the fulfillment of social responsibility. Hypothesis 2 above is verified again.



Variable	Model 1	Model 2	Model 3
	Coefficient (S.E.)	Coefficient (S.E.)	Coefficient (S.E.)
Intercept	-8.921*** (2.042)	-0.354 (10.324)	-8.916*** (2.022)
IC	0.025* (0.013)	0.151*** (0.053)	0.022* (0.013)
FCSR			0.015** (0.007)
L.R&D	0.010 (0.010)	0.013 (0.041)	0.010 (0.010)
L.ROA	2.026*** (0.770)	5.619** (2.196)	1.941** (0.767)
L.LER	-0.056 (0.041)	1.335*** (0.347)	-0.077* (0.041)
L.GROWTH	-0.081 (0.074)	1.464*** (0.227)	-0.103 (0.073)
SHRCR	0.006** (0.003)	0.041** (0.020)	0.005^{*} (0.003)
SUPERVISOR	-0.015 (0.051)	-0.103 (0.310)	-0.013 (0.051)
DUAL	0.104*** (0.039)	0.061 (0.211)	0.103*** (0.039)
L.LNSALARY	0.007 (0.060)	-0.238 (0.418)	0.011 (0.060)
L.DTL	-0.010 (0.008)	0.001 (0.016)	-0.010 (0.008)
LnASSET	0.411*** (0.076)	0.261 (0.427)	0.407*** (0.075)
L.AUDIT	-0.131 (0.179)	0.224 (1.004)	-0.134 (0.174)
STATE	0.125 (0.137)	0.127 (0.222)	0.123 (0.136)
YEAR/ IND	YES	YES	YES
# of obs.	3060	3060	3060
Within_R ²	0.052	0.135	0.056
F_Value	5.46***	15.60***	5.73***

Table 6. Robustness test results before properties are distinguished.

*** Significant at 1%

** Significant at 5%

* Significant at 10%.

Robust standard errors in brackets are clustered at the enterprise level.

https://doi.org/10.1371/journal.pone.0234506.t006

In the robustness test of model 3, IC and FCSR are de-averaged according to the industryannual standard, and the interaction term (IC×FCSR) is added to model 3. Furthermore, model 5 (formula 8) with the interaction term is constructed. The coefficient on IC×FCSR is positive and significant (0.003, p < 0.10), indicating that there is a synergistic effect between internal control and CSR activities, so it is appropriate to analyze the intermediary effect of CSR activities between internal control and innovation performance. Furthermore, model 3 is re-analyzed by regression. As shown in column 4 of Table 6, the coefficient on IC is positive and significant (0.022, p < 0.10). And the coefficient on FCSR is positive and significant (0.015, p < 0.01). Moreover, in accordance with the coefficient on IC in model 1, generally, CSR activities have an intermediary effect in the process of internal control to improve innovation performance. Hypothesis 3 above is verified again. After calculation, the confidence interval corrected of $\beta_1 \times \gamma_2$ for 95% confidence is [0.001, 0.028]. Where, $\beta_1 \times \gamma_2$ is the product of the effect of IC on FCSR and that of FCSR on M_PATENT.

$$M_PATENT_{i,t} = \delta_0 + \delta_1 IC_{i,t} + \delta_2 FCSR_{i,t} + \delta_3 IC_{i,t} \times FCSR_{i,t} + \delta_4 R\&D_{i,t-1} + \delta_5 ROA_{i,t-1} + \delta_6 LER_{i,t-1} + \delta_7 GROWTH_{i,t-1} + \delta_8 SHRCR_{i,t} + \delta_9 SUPERVISOR_{i,t} + \delta_{10} DUAL_{i,t} + \delta_{11} LNSALARY_{i,t-1} + \delta_{12} DTL_{i,t-1} + \delta_{13} LnASSET_{i,t} + \delta_{14} AUDIT_{i,t-1} + \delta_{15} STATE_{i,t} + \delta_{16} \sum_{t} YEAR + \delta_{17} \sum_{t} IND + \varepsilon_{i,t}$$
(8)

For the control variables, in model 1 and model 3, L.ROA, DUAL and LnASSET have



convergent conclusions with the corresponding variables in Table 4 above. In model 2, L. ROA, L.LER, L.GROWTH and SHRCR have convergent conclusions with the corresponding variables in Table 4 above. In addition, in model 1 and model 3, the coefficients on SHRCR are positive and significant (0.006, p < 0.05; 0.005, p < 0.10), implying that the convergence of the interests of major shareholders and the enterprise as a whole will be a favorable factor to improve innovation performance. In model 3, the coefficient on L.LER is negative and significant (-0.077, p < 0.10), suggesting that the higher debt pressure will take a negative impact on innovation performance.

7.2 Robustness tests after properties are distinguished

In order to ensure the reliability of the analysis results, the measurement methods of core indicators are changed again in the robustness tests for the state-owned and non-state-owned samples. For the explained variable in model 1 and model 3, this study takes into account the provision in the article 22 of the "Patent Law of China" that the inventions and utility models for which the patent rights are granted shall be novel, creative and practical. When an enterprise files an application, the patent can only become the real intellectual property of the enterprise after it is officially authorized. Therefore, based on the research of Feng et al. (2017) [79], and Luong et al. (2017) [80], the number of the annual patent authorizations is used as a measure of innovation performance. Where, the number of patents authorized here refers to the total number of patent applications that are filed by an enterprise and are eventually granted in one year. And this variable is symbolized as "GPATENT". In the regression analysis, in order to maintain the dimensional consistency, the natural logarithm of 1 plus the variable is taken and denoted as "LnGPATENT". Moreover, in accordance with the correlation of variables, LnGPATENT is significantly and positively correlated with LnPATENT (0.860, p < 0.10), which also ensures the feasibility of introducing LnGPATENT into model 1 and model 3. Meanwhile, for the explained variable in model 2, the studies of Piotroski et al. (2015) [81] and Li et al. (2019) [18] are referenced, the CSR score (HXCSR) published by "Hexun" is adopted as a measure of CSR performance. The "Hexun" is a subsidiary of China securities market research and design center. "Hexun" score is calculated in accordance the annual reports of listed enterprises and CSR reports published by Chinese stock exchanges, and measures CSR performance from five aspects, including the responsibilities to shareholders, employees, supply chain, environment and public welfare. The total score of CSR performance is weighted in the above five aspects, with a maximum of 100 points. The higher the total score, the higher the level of fulfilling social responsibility, and the higher the enthusiasm to take social responsibility. Then, the corresponding variables in the original model 1-model 3 above are replaced again. And then the state-owned and non-state-owned samples are distinguished for the regression analyses.

On the basis of distinguishing the property attributes, in accordance with the PSM samples, LSDV method is used to analyze model 1—model 3, and the null hypothesis that "the coefficients of all individual dummy variables are zero" is rejected, indicating the existence of individual fixed effect. The robust Hausman tests for model 1—model 3 show that the Sargan-Hansen χ^2 are significant and positive statistically (193.713, 80.572, p = 0.000; 162.191, 74.973, p = 0.000; 195.990, 81.334, p = 0.000), respectively, indicating that the fixed effect model should be adopted. Table 7 reports the robustness test results after the property rights are distinguished.

As shown in Table 7, for model 1, in the state-owned sample, the coefficient on IC is not statistically significant (0.041, p > 0.10). And in the non-state sample, the coefficient on IC is positive and significant (0.058, p < 0.05). Once again, these results show that the positive effect



Variable	Model 1		Model 2		Model 3	
	State-owned	Non-state	State-owned	Non-state	State-owned	Non-state
	Coefficient (S. E.)	Coefficient (S. E.)	Coefficient (S. E.)	Coefficient (S. E.)	Coefficient (S. E.)	Coefficient (S. E.)
Intercept	-6.298** (2.585)	-7.640*** (2.312)	-38.723 (53.472)	-1.344 (37.824)	-6.370** (2.582)	-7.637*** (2.319)
IC	0.041 (0.033)	0.058 ^{**} (0.027)	0.012 (0.900)	1.591** (0.631)	0.041 (0.033)	0.055** (0.028)
HXCSR					-0.002 (0.001)	0.002* (0.001)
L.R&D	0.009 (0.018)	0.038*** (0.011)	0.503 (0.360)	-0.210 (0.244)	0.010 (0.018)	0.038*** (0.010)
L.ROA	0.825(0.754)	0.599 (0.661)	30.992* (16.603)	26.959** (13.165)	0.882 (0.763)	0.549 (0.663)
L.LER	-0.032 (0.049)	-0.085 (0.070)	0.936 (1.001)	1.231 (0.862)	-0.031(0.049)	-0.088 (0.070)
L.GROWTH	-0.011 (0.062)	0.084 (0.059)	3.057 (2.038)	0.344 (1.187)	-0.006 (0.061)	0.083 (0.059)
SHRCR	0.009* (0.005)	-0.0004 (0.003)	-0.003(0.084)	0.132** (0.054)	0.009* (0.005)	-0.001 (0.003)
SUPERVISOR	-0.008 (0.048)	-0.032 (0.069)	0.625 (1.250)	0.068 (1.822)	-0.007 (0.049)	-0.032 (0.069)
DUAL	0.012 (0.070)	0.147^{**} (0.068)	-1.163 (1.953)	1.742 (1.174)	0.010 (0.070)	0.144 ^{**} (0.069)
L.LNSALARY	0.053 (0.077)	0.067 (0.095)	2.828 (1.835)	-0.433 (1.164)	0.058 (0.076)	0.068 (0.095)
L.DTL	-0.001 (0.004)	0.003 (0.007)	-0.243** (0.119)	-0.212 (0.133)	-0.002 (0.004)	0.004 (0.007)
LnASSET	0.302*** (0.091)	0.435*** (0.082)	0.770 (2.190)	0.956 (1.533)	0.304*** (0.091)	0.433*** (0.082)
L.AUDIT	0.137 (0.306)	-0.147 (0.126)	2.760 (3.373)	-5.214 (7.117)	0.142(0.302)	-0.138 (0.131)
YEAR/ IND	YES	YES	YES	YES	YES	YES
# of obs.	1287	1773	1287	1773	1287	1773
Within_R ²	0.138	0.124	0.263	0.135	0.140	0.126
F_Value	6.87***	9.78***	15.26***	10.75***	6.70***	9.43***

Table 7. Robustness test results after property rights are distinguished.

*** Significant at 1%

** Significant at 5%

* Significant at 10%.

Robust standard errors in brackets are clustered at the enterprise level.

https://doi.org/10.1371/journal.pone.0234506.t007

of internal control on technological innovation is restricted by the property attributes of enterprises. This also indicates that state-owned enterprises should strengthen the positive effect of internal control on technological innovation. For model 2, in the non-state sample, the coefficient on IC is positive and significant (1.591, p < 0.05). And in the state-owned sample, the coefficient on IC is not statistically significant (0.012, p > 0.10). This result is slightly different from the corresponding variable in Table 5 above. The possible reason is that "Hexun" gives a higher evaluation on the role of internal control in strengthening innovation performance in non-state-owned enterprises. For model 3, in the state-owned sample, the coefficients on IC and HXCSR are not statistically significant (0.041, p > 0.10; -0.002, p > 0.10). And in the non-state sample, the coefficients on IC and HXCSR are positive and significant (0.055, p < 0.05; 0.002, p < 0.10). Once again, the above results show that the intermediary effect of CSR activities between internal control and innovation performance is only reflected in non-state-owned enterprises. And after calculation, the effect size of the intermediary effect is about 5.49% (i.e. $\beta 1 \times \gamma 2/\alpha 1 = 1.591 \times 0.002/0.058$).



For the control variables, in model 1 and model 3, L.R&D, SHRCR, DUAL and LnASSET have convergent conclusions with the corresponding variables in <u>Table 5</u> above. In model 2, L. ROA, SHRCR and L.DTL have convergent conclusions with the corresponding variables in <u>Table 5</u> above.

8 Conclusions and recommendations

8.1 Conclusions

From the perspective of the effectiveness of internal control, this study analyzes the impact of internal control on enterprise innovation performance and CSR performance. On this basis, this study analyzes the intermediary effect of CSR activities in the process of internal control influencing innovation performance. The conclusions of the study are as follows. The improvement of the effectiveness of internal control has a significant promoting effect on enterprise innovation performance, which is consistent with the research conclusions of Lu et al. (2011) [73] and Chen et al. (2018) [20]. To improve the innovation ability, enterprises should attach importance to the internal system. And the enhancement of the effectiveness of internal control promotes enterprises to significantly improve CSR performance. Enterprises strengthen the construction of their own internal system and safeguard the legitimate rights and interests of stakeholders, which to a large extent reflect the basic objectives of social governance.

Also, on basis of the literature on the relationship between internal control and innovation performance [19, 20], this study introduces the intermediary variable–CSR performance. The results show that the improvement of CSR performance is a positive driving factor for strengthening innovation performance, and CSR activities take an intermediary effect in the process of internal control to improve innovation performance. The further analyses show that the positive effect of internal control on technological innovation is subject to the property attributes. In state-owned enterprises, internal control does not take the significant and positive effect on innovation performance. And the effective internal control promotes enterprises to significantly improve CSR performance. In non-state-owned enterprises, CSR activities play a partial intermediary effect in the process of internal control to improve innovation performance. Such as the provide enterprises are provided enterprises.

In accordance with the existing literature, scholars have expounded the influence of internal control on the risks of technological innovation [15, 16]. Also, scholars have elaborated on the relationship between internal control and CSR activities to a certain extent [17, 18]. Besides, Akisik and Gal (2017) [82] studied the relationship among CSR reports, internal control and financial performance. However, there are few literature on the relationship among internal control, CSR and innovation performance. On basis of the relevant studies of Belloc (2012) [19] and Chen et al. (2018) [20], this study explores deeply the intermediary effect of CSR activities between internal control and innovation performance. This study expounds the mechanism that internal control and CSR influence enterprise innovation performance, and enriches the existing literature on the relationship between internal control of internal control and the relationship between internal control of internal control and CSR influence value for strengthening the construction of internal control, improving the efficiency of innovation output and strengthening CSR practices under the background of economic transformation and upgrading.

8.2 Recommendations

Enterprises should continuously improve the effectiveness of internal control, so as to promote the successful realization of their business objectives and strategic objectives. Within an



industry, the more communication and collaboration between state-owned and non-stateowned enterprises should be encouraged to prevent the possible negative impact of rigid control points, budget and performance appraisal system on innovation activities. In the implementation of internal control, the relationship between institutional control and flexibility should be dynamically balanced, the content and proportion of each control element should be refined, and the formalization of institutional construction and management rigidity should be avoided. The effectiveness of internal control should be strengthened, to strengthen the institutional foundation to enhance the innovation vitality of enterprises, and further improve the efficiency of technological innovation.

In the process of improving innovation performance, enterprises should carefully design rules to balance the private and social benefits of innovators. A normative system of CSR information disclosure is constructed from the perspective of the effectiveness of internal control. The regulators encourage auditors and enterprises to add gradually the relevant contents of "CSR assessment" in the audit reports and self-assessment reports of internal control, respectively, so as to enhance the reliability of CSR information disclosure. Moreover, enterprises ensure the effectiveness of internal control, strengthen the practical performance of social responsibility, and improve the protection of stakeholders' rights and interests. On basis of improving CSR performance, enterprises should establish the good corporate image and deliver the better reputation information to stakeholders, and then obtain and maintain the lasting competitive advantages.

From the long-term perspective of internal control construction, enterprises should make greater efforts in CSR practices. The regulators should promote the joint mechanism of internal control and CSR to protect the legitimate rights and interests of stakeholders. For instance, under the impact of the new coronavirus pneumonia, enterprises should strengthen the institutional responsibility of scientific prevention and control, gradually restore to the normal production and operation, and then promote the economic and social order to return to better normality. Meanwhile, in the process of improving innovation performance through the institutional improvement, the intermediary effect of CSR activities should be promoted. It is necessary to explore the scientific and technological cooperation among those cross-ownership enterprises, identify the opportunities related to CSR strategies, and integrate the R&D behaviors into CSR strategies. Enterprises with different property attributes should work together to establish the R&D centers, make the major technological breakthroughs in R&D activities, improve their innovation performance to reward stakeholders' expectations and demands, and maintain the close ties with all stakeholders.

Supporting information

S1 Dataset. The data set used in this article for discussion and analysis. $(\rm ZIP)$

Author Contributions

Conceptualization: Xiao Li. Data curation: Xiao Li. Formal analysis: Xiao Li. Funding acquisition: Xiao Li. Investigation: Xiao Li. Methodology: Xiao Li.



Project administration: Xiao Li.

Resources: Xiao Li.

Software: Xiao Li.

Supervision: Xiao Li.

Validation: Xiao Li.

Visualization: Xiao Li.

Writing – original draft: Xiao Li.

Writing – review & editing: Xiao Li.

References

- Hao DY, Qi GY, Wang J. Corporate Social Responsibility, Internal Controls, and Stock Price Crash Risk: The Chinese Stock Market. Sustainability 2018; 10, 1675. https://doi.org/10.3390/su10051675
- Bocquet R, Le Bas C, Mothe C, Poussing N. CSR, innovation, and firm performance in sluggish growth contexts: A firm-level empirical analysis. J. Bus. Ethics 2017; 146(1): 241–254. https://doi.org/10.1007/ s10551-015-2959-8
- Yang XD, Peng CC, Yao AL. Management Capability, Internal Control and Corporate Sustainability. *Audit. Res.* 2018; 34(3): 121–128. http://qikan.cqvip.com/Qikan/Article/Detail?id=675401867&from= Qikan_Search_Index
- Zhang HJ. Internal Control, Legal Environments and Corporate Financing Efficiency: Evidence from A-Share Listed Companies in China. J. Shanxi Financ. Econ. Univ. 2019; 39(7): 84–97. <u>https://www.cnki.net/kcms/doi/10.13781/j.cnki.1007-9556.2017.07.007.html</u>
- Xie Z, Kong LX. Executives Overconfidence, Internal Control and Investment Efficiency: Empirical Evidence Based on Chinese A-share Listed Companies. J. Ind. Technol. Econ. 2018; 37(7): 59–66. http://www.wanfangdata.com.cn/details/detail.do?_type=perio&id=gyjsjj201807008
- Yang YW. Measurement of Effect of Chinese Version of SOX on Financial Performance of Listed Companies. J. Shanghai Jiaotong Univ. Philos. Soc. Sci. 2020; 28(1): 58–74. <u>http://kns.cnki.net/kcms/</u> detail/detail.aspx?doi=10.13806/j.cnki.issn1008-7095.2020.01.010
- 7. Zhou ZS, Luo ZY, Zhou XY, Shen Y. Internal Control, Corporate Investment and Corporate Option Value. *Account. Res.* 2017; 38(12): 38–44.
- Song JB, Su ZH, Wang DH. On the Internal Control System with Chinese Characteristic. Account. Res. 2018; 39(9): 11–16.
- Lin C, Lin P, Song FM. Managerial Incentives, CEO Characteristics and Corporate Innovation in China's Private Sector. J. Comp. Econ. 2011; 39(2): 176–190. https://doi.org/10.1016/j.jce.2009.12.001https:// s100.copyright.com/AppDispatchServlet?publisherName=ELS&contentID= S0147596709000894&orderBeanReset=true
- Hirshleifer D, Low A, Teoh SH. Are Overconfident CEOs Better Innovators? J. Financ. 2012, 67(4): 1457–1498. https://doi.org/10.1111/j.1540-6261.2012.01753.x
- Yi J, Wang C, Kafouros M. The Effects of Innovative Capabilities on Exporting: Do Institutional Forces Matter. Int. Bus. Rev. 2013; 22(2): 392–406. https://doi.org/10.1016/j.ibusrev.2012.05.006
- 12. Shen Q, Xie DY. The Sustainability of Global Economic Growth: Beyond All Doubt. *J. Financ. Res.* 2016; 39(2): 86–95.http://qikan.cqvip.com/Qikan/Article/Detail?id=676849230&from=Qikan_Search_ Index
- Zhang ZG, Liang ZG, Yin KG. Research on Corporate Social Responsibility from the Stakeholder Perspective. China Soft Sci. 2012; 27(2): 139–146.
- 14. Song Y. Review and countermeasures on corporate social responsibility under the new normal of economy. *Hum. Resour. Manage.* 2016, 10(6): 74–75.
- Leonce LB, Kenneth ML, Chad JZ. Sarbanes-Oxley and corporate risk-taking. J. Account. Econ. 2010; 49(1): 34–52. https://doi.org/10.1016/j.jacceco.2009.05.001
- Schroeder JH, Shepardson ML. Do SOX 404 Control Audits and Management Assessments Improve Overall Internal Control System Quality? *Account. Rev.* 2015; 91(5): 1513–1541. <u>https://doi.org/10.2308/accr-51360</u>



- Ntim CG, Soobaroyen T. Corporate Governance and Performance in Socially Responsible Corporations: New Empirical Insights from a Neo-Institutional Framework. *Corp. Gov.* 2013; 21(5): 468–494. https://doi.org/10.1111/corg.12026
- Li LY, Wang ZH, Kan LN. Internal Control and Corporate Social Responsibility Performance: Mediating Effect Test Based on Agency Cost. J. Nanjing Audit Univ. 2019; 16(1): 28–36.http://qikan.cqvip.com/ Qikan/Article/Detail?id=7001134952&from=Qikan_Search_Index
- Belloc F. Corporate governance and innovation: A survey. J. Econ. Surv. 2012; 26(5): 835–864. <u>https://doi.org/10.1111/j.1467-6419.2011.00681.x</u>
- Chen H, Na CH, Yutian MZ, Han XF. Internal Control and R&D Subsidy Performance. Manage. World 2018; 34(12): 149–164. https://www.cnki.net/kcms/doi/10.19744/j.cnki.11-1235/f.2018.0040.html
- Eberhart A, Maxwell W, Siddique A. A Reexamination of the Tradeoff between the Future Benefit and Riskiness of R&D Increases. J. Account. Res. 2008; 46(1): 27–52. <u>https://www.jstor.org/stable/ 40058084</u>
- Dougherty D, Hardy C. Sustained Product Innovation in Large, Mature Organizations: Overcoming Innovation-to-organization Problems. Acad. Manage. J. 1996; 39(5): 1120–1153. https://www.jstor.org/ stable/256994
- Verona G. A Resource-based View of Product Development. Acad. Manage. Rev. 1999; 24(1): 132– 142. https://www.jstor.org/stable/259041
- 24. Jensen MC. The Modern Industrial Revolution, Exit, and the Failure of Internal Control Systems. J. *Financ.* 1993; 48 (3): 831–880. https://doi.org/10.1111/j.1745-6622.2010.00260.x
- 25. Kaplan RS, Norton DP. The Balanced Scorecard: Translating Strategy into Action. Boston: Harvard Business Press; 1996.
- Liu XH, Wang AG, Liu XG. Risk Management, Executive Incentives and Innovation Efficiency: An Empirical Analysis Based on China's GEM Listed Companies. *Reform Econ. Syst.* 2018; 36(6): 117– 124.http://gikan.cqvip.com/Qikan/Article/Detail?id=676852306&from=Qikan_Search_Index
- Zhang J, Huang ZZ. Internal Control, Innovation and Firm Performance: Empirical Evidence from Chinese Manufacturing Companies. *Econ. Manage*. 2016; 38(9): 120–134. https://www.cnki.net/kcms/doi/ 10.19616/j.cnki.bmj.2016.09.009.html
- Xu Y, Feng JK, Yang F. Media Coverage, Effectiveness of Internal Control and Innovation Performance. *Collect. Essays Financ. Econ.* 2017; 30(12): 88–96. <u>https://www.cnki.net/kcms/doi/10.13762/j.cnki.cjlc.2017.12.009.html</u>
- Yang DG, Wang JN, Chen LR. Hypercorrection or Trimming? The Role of Internal Control in Corporate Innovation. *Econ. Manage. J.* 2019; 41(8): 113–129. http://kns.cnki.net/kcms/detail/detail.aspx?doi= 10.19616/j.cnki.bmj.2019.08.007
- **30.** Li YM, Shi Q. Can internal control promote the innovation performance of enterprises? *Sci. Res. Manage.* 2019; 40(6): 86–98. http://kns.cnki.net/kcms/detail/detail.aspx?doi=10.19571/j.cnki.1000-2995. 2019.06.009
- Li Y, Zhang C, Liu Y, Li M. Organizational learning, internal control mechanisms, and indigenous innovation: the evidence from China. *IEEE Trans. Eng. Manage*. 2010; 57(1): 63–77. <u>https://ieeexplore.ieee.org/document/5373859</u>
- Shi LP, Liu Q, Wu KJ, Teng Y. FDI Technology Spillover, Knowledge Innovation and Competitive Advantage: Moderating Effect of Absorptive Capacity and Internal Control Mechanism. *R. D. Manage*. 2014; 26(5): 1–13. https://www.cnki.net/kcms/doi/10.13581/j.cnki.rdm.2014.05.001.html
- Li P, Shu W, Tang QQ, Cao J. Can Internal Control Improve the Co-relation between R&D Investment and Firm's Value. *China Account. Rev.* 2015; 13(3): 325–346.
- 34. Jones TM, Wicks AC. Convergent Stakeholder Theory. Acad. Manag. Rev. 1999; 24(5): 206–221. https://doi.org/10.5465/amr.1999.1893929
- Parmar BL, Freeman RE, Harrison JS, Wicks AC, Purnell L, de Colle S. Stakeholder Theory: The State of the Art. Acad. Manag. Ann. 2010; 4(3): 403–445. https://doi.org/10.5465/19416520.2010.495581
- Lonka K, Lindblom-Ylainne S, Maury S. The Effect of Study Strategies on Learning from Text. Learn Instr. 1994; 4(3): 253–271. https://doi.org/10.1016/0959-4752(94)90026-4
- Barnett ML, Salomon RM. Does it pay to be really good? Addressing the shape of the relationship between social and financial performance. *Strateg. Manage. J.* 2012; 33(11): 1304–1320. https://doi. org/10.1002/smj.1980
- Franco S, Caroli MG, Cappa F, Del Chiappa G. Are you good enough? CSR, quality management and corporate financial performance in the hospitality industry. *Int. J. Hosp. Manag.* 2019; 102395. <u>https:// doi.org/10.1016/j.ijhm.2019.102395</u>



- Qin DS. Corporate Social Responsibility, Internal Control and Brand Value. Value Eng. 2019; 38(34): 296–299. http://kns.cnki.net/kcms/detail/detail.aspx?doi=10.14018/j.cnki.cn13-1085/n.2019.34.126
- Hediger W. Welfare and capital—theoretic foundations of corporate social responsibility and corporate sustainability. J. Socio—Econ. 2010; 39(4): 518–526. https://doi.org/10.1016/j.socec.2010.02.001
- 41. Li WA, Dai WT. Corporate Governance, Internal Control, Risk Management: A Framework of Relations: Based on the View of Strategic Management. J. Audit Econ. 2013; 29(4): 3–12.
- Ke B, Lennox CS, Xin QQ. The effect of China's weak institutional environment on the quality of big four audits. Account. Rev. 2015; 90(4): 1591–1619. https://doi.org/10.2308/accr-50943
- Tang XJ. Internal Control, Institutional Environment, and Disclosure Quality of Corporate Social Responsibility. J. Account. Econ. 2016; 30(2): 85–104. https://www.cnki.net/kcms/doi/10.16314/j.cnki. 31-2074/f.2016.02.006.html
- Altamuro J, Beatty A. How Does Internal Control Regulation Affect Financial Reporting? J. Account. Econ. 2010; 49(1): 58–74. https://doi.org/10.1016/j.jacceco.2009.07.002
- Dowdell TD, Kim JC, Klamm BK, Watson MW. Internal Control Reporting and Market Liquidity. *Res. Account. Regul.* 2013; 25(1): 30–40. https://doi.org/10.1016/j.racreg.2012.11.009
- Mitra S, Jaggi B, Hossain M. Internal control weaknesses and accounting conservatism: evidence from the post-Sarbanes-Oxley period. J. Account. Audit. Financ. 2013; 28(2): 152–191. https://doi.org/10. 1177/0148558X13479057
- Du SL, Bhattacharya CB, Sen S. Maximizing Business Returns to Corporate Social Responsibility (CSR): The Role of CSR Communication. Int. J. Manag. Rev. 2010; 26(1): 8–19. <u>https://doi.org/10.1111/j.1468-2370.2009.00276.x</u>
- Aguinis H, Glavas A. What We Know and Don't Know about Corporate Social Responsibility: A Review and Research Agenda. J. Manage. 2012; 38(4): 932–968. <u>https://doi.org/10.1177%</u> 2F0149206311436079
- 49. Padgett RC, Galan JI. The effect of R&D intensity on corporate social responsibility. J. Bus. Ethics 2010; 93(3): 407–418. https://doi.org/10.1007/s10551-009-0230-x
- Luo J, Jia XP. Corporate Social Responsibility (CSR) Initiatives and Firm Innovation: Based on the Social Capital Theory. *R D Manage*. 2017; 29(4): 104–114. https://www.cnki.net/kcms/doi/10.13581/j. cnki.rdm.2017.04.009.html
- Li WX, Jia XP, Liao YH, Liu Y. An Empirical Research of the Influence of Corporate Social Responsibility(CSR) on Firm's Technological Innovation Performance with the Integration of Multiple Perspectives. *Chin. J. Manage.* 2018; 15(2): 237–245.
- Wang W, Zhao X-Z, Chen F-W, Wu C-H, Tsai S, Wang J. The Effect of Corporate Social Responsibility and Public Attention on Innovation Performance: Evidence from High-polluting Industries. *Int. J. Envi*ron. Res. Public Health 2019, 16, 3939. https://doi.org/10.3390/ijerph16203939
- Yang Z, Li JL. Innovation tools or whitewash tools? Reexamination of performance decline and corporate. Studies Sci. Sci. 2020, 38(4): 734–746.http://kns.cnki.net/kcms/detail/detail.aspx?doi=10.16192/j. cnki.1003-2053.2020.04.019
- Garcia R, Calantone R. A critical look at technological innovation typology and innovativeness terminology: A literature review. J. Prod. Innov. Manage. 2002; 19(2):110–132. https://doi.org/10.1016/S0737-6782(01)00132-1
- Li Y, Wei Z, Zhao J, Zhang CL, Liu Y. Ambidextrous organizational learning, environmental munificence and new product performance: Moderating effect of managerial ties in China. *Int. J. Prod. Econ.* 2013; 146(1): 95–105. https://doi.org/10.1016/j.ijpe.2012.11.008
- 56. Zeng P, Li MX, Liu Y. Government Support, Dynamic Model Innovation: Mediation Capabilities, and Business and Moderation Effects. R. D. Manage. 2016; 28(4): 31–38. https://www.cnki.net/kcms/doi/ 10.13581/j.cnki.rdm.2016.04.002.html
- Wang CL, Rodan S, Fruin M, Xu XY. Knowledge Networks, Collaboration Networks, and Exploratory Innovation. Acad. Manage. J. 2014; 57(2): 484–514. https://doi.org/10.5465/amj.2011.0917
- Fang E. Customer participation and the trade-off between new product innovativeness and speed to market. J. Mark. 2008; 72(4): 90–104. https://www.jstor.org/stable/30162301
- Bellamy M, Ghosh S, Hora M. The influence of supply network structure on firm innovation. J. Oper. Manag. 2014; 32 (6): 357–373. https://doi.org/10.1016/j.jom.2014.06.004
- Luo X, Du S. Exploring the Relationship between Corporate Social Responsibility and Firm Innovation. Mark. Lett. 2015; 26(4): 703–714. https://link.springer.com/article/10.1007%2Fs11002-014-9302-5
- Salancik GR, Pfeffer J. A social information processing approach to job attitudes and task design. Adm. Sci. Q. 1978; 23(2): 224–253. https://www.jstor.org/stable/2392563 PMID: 10307892



- 62. Bhattacharya CB, Korschun D, Sen S. Strengthening stakeholder-company relationships through mutually beneficial corporate social responsibility initiatives. J. Bus. Ethics 2009; 85(2): 257-272. https://doi. org/10.1007/s10551-008-9730-3
- Li YZ, Huang W, Liu Y. Job Embeddedness 'Impact on Innovation Performance of High-Tech Enter-63. prise' R&D Staff: Taking the Innovation Capacity as Mediator. Sci. Sci. Manage. S. T. 2014; 35(3): 135-143.
- 64. Tong T, He W, He Z-L, Lu J. Patent Regime Shift and Firm Innovation: Evidence from the Second Amendment to China's Patent Law. Acad. Manage. Proc. 2014; (1): 14174. https://doi.org/10.5465/ ambpp.2014.14174abstract
- Li WJ, Zheng MN. Is it Substantive Innovation or Strategic Innovation? Impact of Macroeconomic Poli-65. cies on Micro-enterprises' Innovation. Econ. Res. J. 2016; 51(4): 60-73. http://gikan.cgvip.com/Qikan/ Article/Detail?id=668574373&from=Qikan Search Index
- Guan JC, Gao X. Exploring the H-index at Patent Level. J. Am. Soc. Inf. Sci. Technol. 2009; 60(1): 35-66. 40. https://doi.org/10.1002/asi.20954
- Berchicci L. Towards and open R&D system: Internal R&D investment, external knowledge acquisition 67. and innovative performance. Res. Policy 2013; 42(1): 117-127. https://doi.org/10.1016/j.respol.2012. 04.017
- Maggitti PG, Smith KG, Katila R. The complex search process of invention. Res. Policy 2013; 42(6): 68. 90-100. https://doi.org/10.1016/i.respol.2012.04.020
- 69. Tian X, Wang TY. Tolerance for Failure and Corporate Innovation. Rev. Financ. Stud. 2014, 27(1): 211-255. https://doi.org/10.1093/rfs/hhr130
- Fan JPH, Wong TJ, Zhang TJ. Politically Connected CEOs, Corporate Governance, and the Post-IPO 70. Performance of China's Partially Privatized Firms. Appl. Corp. Financ. 2014; 26(3): 85-95. https://doi. org/10.1111/jacf.12084
- Li X, Zheng C, Liu G, Sial MS. The Effectiveness of Internal Control and Corporate Social Responsibil-71. ity: Evidence from Chinese Capital Market. Sustainability 2018; 10, 4006. https://doi.org/10.3390/ su10114006
- 72. Hazarika S, Karkoff JM, Nahata R. Internal Corporate Governance, CEO Turnover and Earnings Management. J. Financ. Econ. 2012; 104(1): 44-69. https://doi.org/10.1016/j.jfineco.2011.10.011
- Lu H, Richardson G, Salterio S. Direct and Indirect Effects of Internal Control Weaknesses on Accrual 73 Quality: Evidence from a Unique Canadian Regulatory Setting. Contemp. Account. Res. 2011; 28(2): 675-707. https://doi.org/10.1111/j.1911-3846.2010.01058.x
- 74. Tang KT, Yang YT. The Effectiveness of Internal Control, Fair Value and Earnings Volatility. Financ. Econ. 2016; 41(7): 121-132.
- 75 Zhang XH, Zhu MX, Wang H. Internal Control, Institutional Environment and Enterprise Innovation. China Bus. Mark. 2017, 31(5): 87-95. https://www.cnki.net/kcms/doi/10.14089/j.cnki.cn11-3664/f. 2017.05.011.html
- Jin XR. Yu DJ. Industrial Characteristics. Innovative Efficiency and Regional Economic Growth. J. Zhe-76. jiang Univ. Hum. Soc. Sci. 2010; 40(5): 116–125. http://www.wanfangdata.com.cn/details/detail.do? _type=perio&id=zjdxxb-rwsh201005014
- 77. Abadie A, Imbens GW. Large Sample Properties of Matching Estimators for Average Treatment Effects. Econometrica 2006; 74(1): 235-267. https://doi.org/10.1111/j.1468-0262.2006.00655.x
- 78. Wang XL, Fan G, Yu J. Report on Chinese marketization index by province (2016). Beijing: Social Sciences Academic Press; 2017.
- 79. Feng GF, Liu H, Feng ZZ, Wen J. Does Stock Liquidity Enhance Technological Innovation? J. Financ. Res. 2017; 40(3): 192-206. http://gikan.cgvip.com/Qikan/Article/Detail?id=672093246&from=Qikan_ Search Index
- Luong H, Moshirian F, Nguyen L, Tian X, Zhang B. How Do Foreign Institutional Investors Enhance 80. Firm Innovation? J. Financ. Quant. Anal. 2017; 52(4): 1449-1490. https://doi.org/10.1017/ S0022109017000497
- Piotroski JD, Wong TJ, Zhang T. Political Incentives to Suppress Negative Financial Information: Evi-81. dence from State-controlled Chinese Firms. J. Account. Rev. 2015; 53(2): 405-459. https://doi.org/10. 1111/1475-679X.12071
- Akisik O, Gal G. The impact of corporate social responsibility and internal controls on stakeholders' 82. view of the firm and financial performance. Sustain. Account. Manag. Policy J. 2017; 8 (3): 246-280. https://doi.org/10.1108/SAMPJ-06-2015-0044



Copyright of PLoS ONE is the property of Public Library of Science and its content may not be copied or emailed to multiple sites or posted to a listserv without the copyright holder's express written permission. However, users may print, download, or email articles for individual use.

